The 1988 Joint Protocol Relating to the Application of the Vienna Convention and the Paris Convention – Explanatory Text

IAEA International Law Series No. 5





Front cover: Reproduced from a painting by Anatoly Burykin

THE 1988 JOINT PROTOCOL RELATING TO THE APPLICATION OF THE VIENNA CONVENTION AND THE PARIS CONVENTION — EXPLANATORY TEXT

The following States are Members of the International Atomic Energy Agency:

AFGHANISTAN ALBANIA ALGERIA ANGOLA ARGENTINA ARMENIA AUSTRALIA AUSTRIA AZERBAIJAN BAHRAIN BANGLADESH BELARUS BELGIUM BELIZE BENIN BOLIVIA BOSNIA AND HERZEGOVINA BOTSWANA BRAZIL BULGARIA BURKINA FASO BURUNDI CAMBODIA CAMEROON CANADA CENTRAL AFRICAN REPUBLIC CHAD CHILE CHINA COLOMBIA CONGO COSTA RICA CÔTE D'IVOIRE CROATIA CUBA CYPRUS CZECH REPUBLIC DEMOCRATIC REPUBLIC OF THE CONGO DENMARK DOMINICA DOMINICAN REPUBLIC ECUADOR EGYPT EL SALVADOR ERITREA **ESTONIA ETHIOPIA** FIJI FINLAND FRANCE GABON GEORGIA GERMANY GHANA GREECE

GUATEMALA HAITI HOLY SEE HONDURAS HUNGARY **ICELAND** INDIA **INDONESIA** IRAN, ISLAMIC REPUBLIC OF IRAQ IRELAND ISRAEL ITALY JAMAICA JAPAN JORDAN **KAZAKHSTAN KENYA** KOREA, REPUBLIC OF **KUWAIT KYRGYZSTAN** LAO PEOPLE'S DEMOCRATIC REPUBLIC LATVIA LEBANON LESOTHO LIBERIA LIBYA LIECHTENSTEIN LITHUANIA LUXEMBOURG MADAGASCAR MALAWI MALAYSIA MALI MALTA MARSHALL ISLANDS MAURITANIA MAURITIUS MEXICO MONACO MONGOLIA MONTENEGRO MOROCCO MOZAMBIQUE MYANMAR NAMIBIA NEPAL NETHERLANDS NEW ZEALAND NICARAGUA NIGER NIGERIA NORWAY OMAN PAKISTAN PALAU

PANAMA PAPUA NEW GUINEA PARAGUAY PERU PHILIPPINES POLAND PORTUGAL OATAR REPUBLIC OF MOLDOVA ROMANIA RUSSIAN FEDERATION RWANDA SAUDI ARABIA SENEGAL SERBIA **SEYCHELLES** SIERRA LEONE SINGAPORE **SLOVAKIA SLOVENIA** SOUTH AFRICA SPAIN SRI LANKA **SUDAN SWAZILAND** SWEDEN SWITZERLAND SYRIAN ARAB REPUBLIC TAJIKISTAN THAILAND THE FORMER YUGOSLAV **REPUBLIC OF MACEDONIA** TOGO TRINIDAD AND TOBAGO TUNISIA TURKEY UGANDA UKRAINE UNITED ARAB EMIRATES UNITED KINGDOM OF GREAT BRITAIN AND NORTHERN IRELAND UNITED REPUBLIC OF TANZANIA UNITED STATES OF AMERICA URUGUAY **UZBEKISTAN** VENEZUELA VIETNAM YEMEN ZAMBIA ZIMBABWE

The Agency's Statute was approved on 23 October 1956 by the Conference on the Statute of the IAEA held at United Nations Headquarters, New York; it entered into force on 29 July 1957. The Headquarters of the Agency are situated in Vienna. Its principal objective is "to accelerate and enlarge the contribution of atomic energy to peace, health and prosperity throughout the world".

IAEA INTERNATIONAL LAW SERIES No. 5

THE 1988 JOINT PROTOCOL RELATING TO THE APPLICATION OF THE VIENNA CONVENTION AND THE PARIS CONVENTION — EXPLANATORY TEXT

INTERNATIONAL ATOMIC ENERGY AGENCY VIENNA, 2013

COPYRIGHT NOTICE

All IAEA scientific and technical publications are protected by the terms of the Universal Copyright Convention as adopted in 1952 (Berne) and as revised in 1972 (Paris). The copyright has since been extended by the World Intellectual Property Organization (Geneva) to include electronic and virtual intellectual property. Permission to use whole or parts of texts contained in IAEA publications in printed or electronic form must be obtained and is usually subject to royalty agreements. Proposals for non-commercial reproductions and translations are welcomed and considered on a case-by-case basis. Enquiries should be addressed to the IAEA Publishing Section at:

Marketing and Sales Unit, Publishing Section International Atomic Energy Agency Vienna International Centre PO Box 100 1400 Vienna, Austria fax: +43 1 2600 29302 tel.: +43 1 2600 22417 email: sales.publications@iaea.org http://www.iaea.org/books

© IAEA, 2013

Printed by the IAEA in Austria April 2013 STI/PUB/1593

IAEA Library Cataloguing in Publication Data

The 1988 joint protocol relating to the application of the Vienna Convention and the Paris Convention : explanatory text. — Vienna : International Atomic Energy Agency, 2013.

p. ; 28 cm. — (IAEA international law series, ISSN 1991–2366 ; no. 5) STI/PUB/1593

ISBN 978-92-0-139410-1

Includes bibliographical references.

1. Radioactive substances — Law and legislation. 2. Liability for nuclear damages. I. International Atomic Energy Agency. II. Series.

IAEAL

13-00811

FOREWORD

by Yukiya Amano Director General

At the International Conference on the Safety of Transport of Radioactive Material, held in Vienna in July 2003, it was noted that there are a number of liability conventions to which many States are Parties but to which many others are not, and that the provisions of the liability conventions and the relationships between them are not simple to understand. It was suggested, therefore, that an explanatory text for these instruments be prepared to assist in developing a common understanding of the complex legal issues involved and thereby to promote adherence to these instruments.

In view of this suggestion, and in order to foster an effective, global nuclear liability regime, the International Expert Group on Nuclear Liability (INLEX) was established in September 2003. Since its establishment, INLEX has held several meetings, as a result of which it, inter alia, finalized explanatory texts on the nuclear liability instruments adopted under IAEA auspices in 1997, namely, the Protocol to Amend the Vienna Convention on Civil Liability for Nuclear Damage and the Convention on Supplementary Compensation for Nuclear Damage. These explanatory texts have since been published as IAEA International Law Series No. 3.

Thereafter, INLEX also developed and finalized the present explanatory text on the 1988 Joint Protocol Relating to the Application of the Vienna Convention and the Paris Convention. This publication complements IAEA International Law Series No. 3 and comprises the explanatory text on the Joint Protocol and the text of the Joint Protocol as adopted by the Conference on the Relationship between the Paris Convention and the Vienna Convention, which met at IAEA Headquarters in Vienna in September 1988 under the joint auspices of the IAEA and the OECD Nuclear Energy Agency. It is hoped that this publication will further increase the awareness of nuclear liability as an important aspect of nuclear law.

ACKNOWLEDGEMENTS

Appreciation is expressed to the members of INLEX and, in particular, to A. Gioia of the Faculty of Law of the University of Modena and Reggio Emilia, Italy, for the valuable work they have done in developing the text.

EDITORIAL NOTE

Although great care has been taken to maintain the accuracy of information contained in this publication, neither the IAEA nor its Member States assume any responsibility for consequences which may arise from its use.

The use of particular designations of countries or territories does not imply any judgement by the publisher, the IAEA, as to the legal status of such countries or territories, of their authorities and institutions or of the delimitation of their boundaries.

The mention of names of specific companies or products (whether or not indicated as registered) does not imply any intention to infringe proprietary rights, nor should it be construed as an endorsement or recommendation on the part of the IAEA.

Some of the material in this publication was reproduced by scanning original paper copies. The IAEA assumes no responsibility for the accuracy of the text created by this method.

CONTENTS

1.	INTRODUCTION		1
		The existence of two distinct international treaty regimes on civil liability for nuclear damage The problems created by the absence of treaty relations between the Contracting Parties	1
		to the Paris Convention and the Contracting Parties to the Vienna Convention	2
		The different options envisaged in order to create a treaty link between the Contracting Parties to the Paris Convention and the Contracting Parties to the Vienna Convention	
2.		VERAL ASPECTS	
	2.1.	The legal nature of the Joint Protocol and its relation to the Vienna Convention and	
		the Paris Convention	11
		the Vienna Convention and the Paris Convention	
	2.2.	The object and purpose of the Joint Protocol as expressed in its Preamble	13
3.	OPE	RATIVE PROVISIONS	15
	3.1.	The mutual extension of the operator's liability under either Convention	
	3.2.	to damage suffered in the territory of Contracting Parties to the other Convention (Article II) The elimination of the distinction between Contracting Parties and non-Contracting States	15
	3.2.	as regards the application of the operative provisions of either Convention (Article IV)	17
		3.2.1. The determination of the liable operator.	
		3.2.2. The determination of the State whose courts have jurisdiction	
	3.3.	3.2.3. The amount of compensation to be paid	
4.		AL CLAUSES AND TRANSITIONAL ISSUES.	
	41	Depositary functions and authentic texts	31
	4.2.		
		Transitional issues relating to potential conflicts of treaty obligations	
		OTOCOL RELATING TO THE APPLICATION OF THE VIENNA CONVENTION	
A	ND T	HE PARIS CONVENTION	35

1. INTRODUCTION

1.1. THE EXISTENCE OF TWO DISTINCT INTERNATIONAL TREATY REGIMES ON CIVIL LIABILITY FOR NUCLEAR DAMAGE

When the Vienna Convention on Civil Liability for Nuclear Damage (the Vienna Convention) was adopted under the auspices of the International Atomic Energy Agency (IAEA) on 21 May 1963, another international treaty relating to the same subject matter already existed, namely, the Paris Convention on Third Party Liability in the Field of Nuclear Energy (the Paris Convention). The Paris Convention had been adopted on 29 July 1960 under the auspices of the then Organisation for European Economic Co-operation (OEEC), which was later reconstituted as the Organisation for Economic Co-operation and Development (OECD), but it had not yet entered into force at the time when the Vienna Convention was adopted.

Both the Paris Convention and the Vienna Convention purported to create, in the national law of their respective Contracting Parties, a special legal regime for nuclear liability derogating from the otherwise applicable rules governing third party liability. In fact, it was felt at the time that the problems of civil liability for damage caused by incidents at land based nuclear installations, and in the course of transport of nuclear material thereto or therefrom, called for special provisions in national law in order to ensure prompt and adequate compensation for nuclear damage without at the same time exposing the nuclear industry to excessive burdens. In addition, it was felt that the special nature of nuclear hazards, and the possibility that a nuclear incident might cause damage of an extreme magnitude and involve the nationals of more than one country, called for some degree of harmonization of national laws through the adoption of international agreements.¹

In addition to sharing this fundamental purpose, both the Paris Convention and the Vienna Convention laid down very similar nuclear liability rules based on the same general principles: "absolute" liability, that is, liability without fault; exclusive liability of the operator of the nuclear installation concerned; obligation for the operator to cover his liability by insurance or other financial security; limitation of the amount of the operator's liability and/or limitation of liability cover; limitation of the operator's liability in time; and non-discrimination as regards the victims of a nuclear incident. Moreover, both the Paris Convention and the Vienna Convention laid down identical rules on jurisdiction, designed to establish a single competent forum to deal with all actions for compensation arising thereunder, and provided for the recognition and enforcement of final judgements entered by the competent court.²

Despite these very similar rules based on the same basic principles, a number of differences between the Paris Convention and the Vienna Convention remained. To reduce these differences, the Paris Convention was first amended, even before its entry into force, by the Additional Protocol of 28 January 1964; however, after the Paris Convention entered into force (on 1 April 1968), it was further amended by the Additional Protocol of 16 November 1982, which added new differences between the two Conventions. Similarly, when the Protocol to Amend the Vienna Convention on Liability for Nuclear Damage (the Vienna Protocol) was adopted on 12 September 1997, although the need to harmonize the provisions of the Vienna Convention with those of the Paris Convention was taken into account, significant differences were added.³ The differences between the Paris Convention and the Vienna Protocol will be

¹ For more details on the origin of the international civil liability regime and, in particular, of the Vienna Convention, see *The 1997 Vienna Convention on Civil Liability for Nuclear Damage and the 1997 Convention on Supplementary Compensation for Nuclear Damage — Explanatory Texts*, IAEA International Law Series No. 3, IAEA, Vienna (2007), Section 1.

² For more details on the general principles of nuclear liability as reflected, in particular, in the Vienna Convention, see IAEA International Law Series No. 3, Sections 1 and 2.

³ For more details on the differences between the original Vienna Convention and the Vienna Convention as amended by the 1997 Vienna Protocol, see IAEA International Law Series No. 3, Section 2. Under Article 18 of the Vienna Protocol, the Vienna Convention as amended by the 1997 Vienna Protocol may be referred to as the 1997 Vienna Convention, as opposed to the 1963 Vienna Convention. Unless otherwise specified (e.g. 1963 Vienna Convention or 1997 Vienna Convention), all references to the Vienna Convention in this explanatory text are intended to refer to both the 1963 Vienna Convention and the 1997 Vienna Convention.

reduced, but not eliminated, once the Paris Convention is further amended by the Protocol of 12 February 2004, which is not yet in force.

The main remaining difference between the Paris Convention and the Vienna Convention (both in its 1963 version and as amended by the 1997 Vienna Protocol) is the 'regional' character of the Paris Convention, which is only open to OECD Member or Associated States (Article 21), versus the potentially global character of the Vienna Convention: the 1963 Vienna Convention is open to all Member States of the United Nations, its specialized agencies or the IAEA (Article XXIV), and the 1997 Vienna Protocol is open to all States (Article 20). There are, in addition, a number of substantive differences between the legal regimes laid down by the existing conventions, in particular, the different limits envisaged for the amount of the operator's liability (Article 7 of the Paris Convention and Article V of the Vienna Convention).⁴ Another difference is the possibility for a Contracting Party to subject the transit of nuclear substances through its territory to the condition that the maximum amount of liability of the foreign operator concerned be increased to the maximum amount of liability of operators of nuclear installations situated in its territory, which is envisaged by the Paris Convention (Article 7(e)) but not by the Vienna Convention.⁵

There are, in addition, other differences, such as those concerning the conditions for the acquisition of a right of subrogation in cases where persons other than the operator of a nuclear installation have paid compensation for damage caused by a nuclear incident (Article 6(d) and (e) of the Paris Convention and Article IX.2(a) of the Vienna Convention). However, there seems to be no need to examine the substantive differences between the Paris Convention and the Vienna Convention in further detail, at least at this stage⁶, since the problems that the 1988 Joint Protocol Relating to the Application of the Vienna Convention and the Paris Convention (the Joint Protocol) is intended to solve are, to a large extent, independent of those differences. The problems that the Joint Protocol was intended to solve derive from the very existence of distinct treaty regimes that, though largely similar in content, have different Contracting Parties.

1.2. THE PROBLEMS CREATED BY THE ABSENCE OF TREATY RELATIONS BETWEEN THE CONTRACTING PARTIES TO THE PARIS CONVENTION AND THE CONTRACTING PARTIES TO THE VIENNA CONVENTION

The first problem originating from the absence of treaty relations between the Contracting Parties to the two different Conventions relates to the so-called geographical scope of the nuclear liability regime, inasmuch as the Convention under which the operator is liable may not apply to nuclear damage suffered in the territory of the Contracting Parties to a different Convention. The Paris Convention, as at present in force, applies neither to nuclear incidents occurring in non-Contracting States nor to nuclear damage suffered in such States, unless otherwise provided by the legislation of the Contracting Party within whose territory the liable operator's installation is situated (hereinafter referred to as the 'Installation State') (Article 2); although the 1963 Vienna Convention is silent as to its territorial scope, it is often interpreted in the same way.⁷ The 1997 Vienna Convention applies in principle to nuclear damage "wherever suffered", but then allows the legislation of the Installation State to exclude damage suffered in a nuclear

⁴ In addition, the Vienna Convention envisages a minimum limit (Article V), thus not excluding the possibility to provide for unlimited liability, whereas the Paris Convention envisages a maximum limit (Article 7). However, when the 2004 Protocol to amend the Paris Convention enters into force, the liability limit envisaged in the Paris Convention will also be a minimum limit. On the other hand, both the Vienna Convention and the Paris Convention provide that the operator's obligation to cover liability by insurance or other financial security is limited to a given amount that is specified by the State in which the nuclear installation is situated in the case of the Vienna Convention (Article VII) and by the Convention itself in the case of the Paris Convention (Article 10).

⁵ As a result of the negotiations that led to the adoption of both the Vienna Protocol and the Convention on Supplementary Compensation (CSC) in 1997, a provision similar to the one in Article 7(e) of the Paris Convention was inserted in the Annex to the CSC, which applies to States party to neither the Paris Convention nor the Vienna Convention, but not in the Vienna Convention as amended by the Vienna Protocol; see IAEA International Law Series No. 3, Section 3.3.2.

⁶ See the discussion in Section 3.2.

⁷ For more details as to the territorial scope of application of the Paris Convention and the Vienna Convention, see Section 3.1.

non-Contracting State not affording "equivalent reciprocal benefits" (Article I A); once the 2004 Protocol enters into force, a similar provision will be inserted in the Paris Convention as well (Article 2, as amended by the 2004 Protocol).

The second problem originating from the absence of treaty relations between the Contracting Parties to the two different Conventions relates to the determination of the operator liable in cases where nuclear material is being transported between the operators of different nuclear installations.⁸ Although both Conventions lay down identical rules in this respect (Article 4 of the Paris Convention and Article II.1 of the Vienna Convention), these rules are different depending on whether the transport takes place between persons situated within the territories of Contracting Parties or between one person situated in the territory of a Contracting Party and another person situated in the territory of a non-Contracting State. The general rule under all nuclear liability Conventions is that, in the case of transport of nuclear material between the operators of two nuclear installations, the sending operator is liable until the receiving operator has assumed liability pursuant to the express terms of a written contract or, in the absence of such contract, until the receiving operator has taken charge of the material involved. However, this general rule only applies if both operators are within the territory of Contracting Parties to the same Convention: if the nuclear material is sent to a person in a non-Contracting State (including a Contracting Party to a different Convention), the sending operator remains liable until the nuclear material has been unloaded from the means of transport by which it arrived in the territory of that State; conversely, if the nuclear material is sent from a person in a non-Contracting State (including a Contracting Party to a different Convention), liability is imposed upon the receiving operator from the moment the material has been loaded on the means of transport. Thus, if a nuclear incident were to occur in the course of transport of nuclear materials between operators of nuclear installations situated in the territories of Contracting Parties to different nuclear liability Conventions, inasmuch as the applicable rules are those relating to transport to or from persons in non-Contracting States, and therefore the transfer of liability between the sending and the receiving operator cannot take place on the basis of a written contract between them or when one of them has taken charge of the materials involved, both operators may be held liable, each under the applicable Convention, for the entire time the nuclear materials are on the means of transport. One consequence of this situation would be the need for both operators to conclude insurance contracts in order to cover their respective liability for the damage caused by the same nuclear incident.

An additional and related problem originating from the absence of treaty relations between the Contracting Parties to different Conventions relates to the determination of the State whose courts have jurisdiction in the event of a nuclear incident in the course of transport of nuclear material.⁹ The general rule under all nuclear liability Conventions is that jurisdiction lies with the courts of the Contracting Party within whose territory the nuclear incident occurred (hereinafter referred to as the 'Incident State'); however, jurisdiction lies with the courts of the Installation State if the nuclear incident occurred outside the territory of a Contracting Party (Article 13 of the Paris Convention and Article XI of the Vienna Convention).¹⁰ In the case of a nuclear incident occurring in the course of transport of nuclear material between operators situated in Contracting Parties to different nuclear liability Conventions, these provisions may lead to a situation where the courts of both States have jurisdiction in respect of the same nuclear incident, each of them on the basis of the applicable Convention: since each of these States is to be considered as a non-Contracting State vis-à-vis the other, both the sending and the receiving operators may be held liable, each of them under the applicable

⁸ In the case of a nuclear incident at a nuclear installation, there would be no specific problem, since the operator of that installation would be liable under the applicable Convention and the other Convention would not apply.

⁹ In the case of a nuclear incident at a nuclear installation, there would be no specific problem, since only the Convention to which the Installation State is a Contracting Party would apply and the courts of that same State, which is at the same time the Installation State and the Contracting Party within whose territory the nuclear incident occurred (the 'Incident State'), would have exclusive jurisdiction thereunder.

¹⁰ In the case of incidents occurring in the course of maritime transport, the term "territory" under all Conventions is deemed to include maritime areas subject to the coastal State's full sovereignty, such as internal and territorial waters, but not other maritime areas subject to more limited sovereign rights of the coastal State or otherwise subject to its jurisdiction for specific purposes, such as the exclusive economic zone. One of the most notable features of the 1997 Vienna Convention is the fact that, in the case of incidents occurring in the exclusive economic zone of a Contracting Party, jurisdiction lies with the courts of that State rather than with the courts of the Installation State (Article XI). Once the 2004 Protocol enters into force, a similar rule will be inserted in Article 13 of the Paris Convention.

Convention; in addition, the courts in both States, both being the Installation State under the applicable Convention, would have jurisdiction. At the same time, neither of these States would be under a specific treaty obligation to ensure that final judgements entered by the competent court in the other State are recognized and enforced within its territory (Article 13(d) of the Paris Convention and Article XII of the Vienna Convention).

1.3. THE DIFFERENT OPTIONS ENVISAGED IN ORDER TO CREATE A TREATY LINK BETWEEN THE CONTRACTING PARTIES TO THE PARIS CONVENTION AND THE CONTRACTING PARTIES TO THE VIENNA CONVENTION

The issue of the relationship between the Paris Convention and the Vienna Convention was first raised during the 1963 International Conference on Civil Liability for Nuclear Damage, which resulted in the adoption of the Vienna Convention.¹¹ As a result, the Conference agreed to insert two articles in the Vienna Convention. The first one (Article XVI) provides that "No person shall be entitled to recover compensation under this Convention to the extent that he has recovered compensation in respect of the same nuclear damage under another international convention on civil liability in the field of nuclear energy". Pursuant to the second one (Article XVII), the Vienna Convention "shall not, as between the Parties to them, affect the application of any international agreements or international conventions on civil liability in the field of nuclear energy in force, or open for signature, ratification or accession at the date on which this Convention is opened for signature." At the time these provisions were inserted in the Vienna Convention, the only international convention Supplementary to the Paris Convention (the Brussels Convention), which, however, does not itself lay down the rules of nuclear liability and only increases the amount of compensation available for nuclear damage.

Although the Paris Convention was not yet in force at the time (nor, indeed, was the Brussels Convention), one of the reasons why these provisions were inserted in the Vienna Convention was clearly the desire to allow the Contracting Parties to the Paris Convention to become Contracting Parties to the Vienna Convention without at the same time having to terminate the former Convention might at some stage become Contracting Parties to the Vienna Convention as well was also at the basis of the first revision of the Paris Convention by way of the 1964 Additional Protocol, which, as is pointed out in Section 1.1, was intended to harmonize the provisions of the Paris Conventions with those of the Vienna Convention. In fact, the Preamble of the 1964 Additional Protocol made it clear that the desire of the Signatories "of ensuring that as far as possible there are no conflicts between the two Conventions" was meant to enable them "to become parties to both Conventions if they so decide". That possibility was, however, soon made unlikely by the widespread conviction that the ratification of both Conventions might lead to a number of conflicts of treaty obligations for the interested States, a possibility that was evoked during a 1968 symposium jointly organized by the IAEA and the then European Nuclear Energy Agency (now the OECD Nuclear Energy Agency (NEA)), a specialized OECD agency with specific competence in the nuclear field.¹²

As a result, a number of alternative options were examined in order to create a treaty link between the Contracting Parties to either Convention. At its very first session, in 1964, the IAEA Standing Committee on Civil Liability for Nuclear Damage (IAEA SCNL), a body created by the IAEA Board of Governors in 1963 to review and advise on

¹¹ See Civil Liability for Nuclear Damage. Official Records of the International Conference on Civil Liability for Nuclear Damage held by the International Atomic Energy Agency in Vienna, 29 April–19 May 1963, Legal Series No. 2, IAEA, Vienna (1964), in particular pp. 381–384.

¹² Nordenson, U.K., "Legal conflicts arising from the simultaneous application of the Paris and Vienna Conventions with regard to nuclear incidents in the course of carriage of nuclear substances", *Third Party Liability and Insurance in the Field of Maritime Carriage of Nuclear Substances* (Proc. Symp., Monaco, 1968), OECD/NEA, Paris (1969) pp. 427 ff.

problems relating thereto, briefly discussed the issue of the relationship between the Vienna Convention and "regional conventions" on the basis of a letter by Norway.¹³ However, already at its second session, in 1967, the IAEA SCNL decided not to discuss the issue further, and instead asked the IAEA Secretariat, together with Norway, to study it.¹⁴ Discussions resumed in October 1972 within the OECD/NEA Group of Governmental Experts on Third Party Liability in the Field of Nuclear Energy (hereinafter referred to as the "NEA Group"), which invited the NEA Secretariat to prepare a working document, in collaboration with the IAEA Secretariat, "analysing in detail the different formulas that might resolve [the] problem" of the relationship between the Paris and Vienna Conventions, and set up a Working Group (hereinafter referred to as the "NEA WG") in order to examine that document.¹⁵ A first working document on "Possible Solutions to the Problems Concerning the Relationship between the Paris and Vienna Conventions" (hereinafter referred to as "WD I") was, therefore, produced by the NEA and IAEA Secretariats on 17 January 1973; this document, which outlined four alternative options, was discussed on 1–2 February 1973 by the NEA WG, which asked for its revision.¹⁶ As a result, a second working document, titled "Possible Solutions to the Problem of the Relationship between the Paris and the Vienna Conventions" (hereinafter referred to as "WD II"), outlining five (rather than four) alternative options was produced by the two Secretariats on 16 March 1973.¹⁷

Of the five solutions outlined in WD II, the first three were based on the idea that the two existing Conventions should be replaced by a single treaty instrument. These solutions were included in Category I, as follows: (I.A) to make the Paris Convention (which had come into force on 1 April 1968) freely open to all States, but on the clear understanding that the Vienna Convention would not come into force; (I.B) to ask the Contracting Parties to the Paris Convention to terminate that Convention and to ratify the Vienna Convention; and (I.C) to adopt an entirely new convention coalescing the texts of the two existing Conventions and intended to replace both. The remaining two solutions were based on the different idea that the two existing Conventions would continue in being (once the Vienna Convention entered into force), and that some other way to create a treaty link between the Contracting Parties to either should be found. These solutions were included in Category II, as follows: (II.A) ratification of the Vienna Convention by the Contracting Parties to the Paris Convention without terminating the latter Convention; and (II.B) creation of a "bridge" between the two Conventions by means of a similar protocol to each Convention or a joint protocol to both, so that the Parties to either Convention would be treated "as if they were Parties" to the other.

WD II was first discussed within the NEA Group in April 1973.¹⁸ As far as the solutions in Category I were concerned, "It appeared fairly rapidly that neither of the first two solutions were likely to be considered by the Experts", even though "certain Experts...expressed interest in the solution consisting of opening the Paris Convention to countries outside [the] OECD, without this resulting in impeding the entry into force of the Vienna Convention". As for the third solution, "most of the Experts considered that the practical difficulties in implementing it, as well as the risk that the previous Conventions might still be maintained and the principles of nuclear third party liability questioned, were too great compared to its advantages". The experts then turned to the solutions in Category II, and the discussion showed "a certain preference...for solution II.A without, however, solution II.B being set aside". Consequently, it was

¹³ IAEA document CN-12/SC/9, paras 10–12. The Committee reached the following conclusions:

[&]quot;The Committee recognizes that it would be for the States concerned to decide whether they should establish regional conventions on civil liability for nuclear damage and, if so, whether such conventions should be independent of, or supplementary to, the Vienna Convention. Nevertheless, the Committee unanimously expressed the view that any such regional convention should not conflict or compete with the Vienna Convention. A majority of the members of the Committee were of the opinion that it would be preferable if regional conventions were supplementary to the Vienna Convention, while some members felt that regional conventions might be independent."

¹⁴ IAEA document CN-12/SC/14, para. 2.

¹⁵ OECD-NEA document SEN (72)22, paras 16–17.

¹⁶ WD I was never distributed as an official document, and there appears to be no official report of the NEA WG meeting of 1–2 February 1973. However, the meeting and its outcome are referred to in OECD-NEA document SEN(73)6, and a List of Participants is attached thereto as Annex I. The NEA WG consisted of representatives of France, Italy, the Netherlands, Portugal, Sweden, Switzerland and the United Kingdom.

¹⁷ OECD-NEA document SEN(73)6, Annex II.

¹⁸ OECD-NEA document SEN(73)10.

agreed that "these two solutions should be studied further, priority being given to solution II.A," and the NEA and IAEA Secretariats were requested to "prepare jointly a working document concerning the practical consequences of implementing solution II.A and the way remaining conflicts could be settled, for discussion at the next meeting."

Solution II.A in WD II — that is, ratification of the Vienna Convention by the Parties to the Paris Convention — was in fact the option that had already been envisaged in the Preamble of the 1964 Additional Protocol to the Paris Convention and that had soon been discarded in view of the widespread opinion that it would lead to conflicts of treaty obligations. This probably explains why it had not been proposed as a solution in WD I, which only envisaged four options. Its reappearance in WD II, as solution II.A, was due to a specific request by the Swedish representative within the NEA WG when WD I was discussed in February 1973,¹⁹ and the "preference" for that solution at the NEA Group meeting in April 1973 was mainly expressed by the experts from Denmark, Sweden and Norway.²⁰ However, despite the priority given to the study of solution II.A, when the document concerning the practical consequences of implementing that solution, prepared by the NEA and IAEA Secretariats,²¹ was in fact discussed by the NEA Group in October 1973, the Chairman's consultations quickly "revealed that a certain number of countries refused to envisage ratification of the Vienna Convention, a fact depriving this solution of all chances of success". Consequently, the experts agreed to study the remaining solution, namely, solution II.B, "which provided for the adoption of a Protocol for the settling of conflicts between the two Conventions."²²

1.4. THE DRAFTING HISTORY OF THE JOINT PROTOCOL

As is pointed out in Section 1.3, at its October 1973 meeting the NEA Group decided to concentrate its efforts on the creation of a "bridge" between the Paris Convention and the Vienna Convention by means of a similar protocol to each or a joint protocol to both (solution II.B in WD II), this being the only remaining solution after none of the other possible solutions concerning the relationship between the two Conventions had met with sufficient agreement. Consequently, a first draft instrument (hereinafter referred to as the "1974 Draft Joint Protocol")²³ and a brief explanatory note thereon (hereinafter referred to as the "1974 Explanatory Note")²⁴ were produced by the NEA and IAEA Secretariats. It was accepted that the 1974 Draft Joint Protocol, together with the other options originally envisaged in WD II, would first be considered within a "Restricted Working Group" of the IAEA SCNL, which would meet in Vienna from 7 to 9 May 1974.²⁵ Since this Restricted Working Group, in turn, recommended to the IAEA SCNL that it pursue in detail the solution of a protocol establishing a relationship between the Vienna Convention and the Paris Convention,²⁶ the 1974 Draft Joint Protocol was then more formally discussed within the NEA Group in June 1974 and again in March 1975.

In June 1974, no clear conclusion on the protocol solution was reached by the NEA Group and, inter alia, the question was raised as to whether a satisfactory solution could be achieved by two other means that until then had not been discussed in detail, the first consisting of an appropriate conflict rule rather than a "bridge" between the two Conventions (the Netherlands), and the second consisting of a unilateral extension of the territorial scope of application of the Paris Convention, pursuant to its Article 2, to nuclear incidents occurring and damage suffered in the territory of non-Contracting States (Germany and the Netherlands). Consequently, the NEA and IAEA Secretariats were requested to submit a more detailed document on the subject in advance of the next meeting of the NEA Group.²⁷ In accordance

¹⁹ OECD-NEA document SEN(73)6.

²⁰ OECD-NEA document SEN(73)10.

²¹ OECD-NEA document SEN(73)14.

²² OECD-NEA document SEN(74)2.

²³ OECD-NEA document SEN(74)7, Annex II. The text of the 1974 Draft Joint Protocol was also later reproduced in an annex to IAEA document N5.TC-462.6/2.

²⁴ OECD-NEA document SEN(74)7, Annex I.

²⁵ OECD-NEA document SEN(74)7.

²⁶ No official record of the meeting of this working group survives. However, the conclusion was reported by the IAEA representative at the next meeting of the NEA Group in June 1974: see OECD-NEA document SEN(74)14.

²⁷ OECD-NEA document SEN(74)14.

with that request, a document entitled "Relationship between the Paris and Vienna Conventions" (hereinafter referred to as "WD III")²⁸ was prepared in January 1975 by the two Secretariats in order to supplement the 1974 Draft Joint Protocol and the 1974 Explanatory Note.

WD III outlined a number of cases involving the simultaneous application of both the Paris Convention and the Vienna Convention and, for each such case, examined whether problems raised thereby could be solved by the solutions discussed at the previous NEA Group meeting, that is, a unilateral extension of the Paris Convention, a conflict rule or the proposed protocol. The conclusions arrived at in WD III made it clear that the protocol solution "would combine the advantages of the two foregoing solutions without having their drawbacks" and, in addition, would "offer an opportunity to realize the idea of achieving a world-wide system of nuclear third party liability in which all countries would participate irrespective of their differing constitutional and social systems"; and finally, that "A better protection of victims of a possible nuclear incident would go hand in hand with an improvement of insurance conditions and the furtherance of international collaboration in the field of nuclear energy."²⁹ While recognizing that the protocol solution might cause some difficulties of its own in the transitional period (i.e. the period between the date of its entry into force and the date on which all Parties to both Conventions are also Parties to the Protocol), WD III concluded that "the Protocol would appear to be the most effective solution to the problems discussed above".³⁰

WD III was discussed within the NEA Group in March 1975. After extensive discussion of all of the cases presented therein, the experts, in the Chairman's words:

"had not found arguments militating against the legal conclusion of the Secretariats that the conflict rule and the unilateral extension of the Paris Convention would not resolve all problems of the relationship between the two Conventions as would the Joint Protocol. There had been, however, a number of reservations to the latter solution, in particular as regards the question of amendment of the Conventions and the applicability of Article 7(e) of the Paris Convention."³¹

The question of the possible effects of the adoption of a joint protocol on the application of the Brussels Convention also raised concerns.³² It was eventually agreed that the IAEA Secretariat would submit WD III to the IAEA SCNL, together with the experts' observations and comments during the meeting.³³ However, the IAEA SCNL took no action at this stage, probably because the Vienna Convention was not yet in force and, consequently, the question of the relationship between the Vienna Convention and the Paris Convention was not considered urgent: the question of the relationship between the two Conventions was not even placed on the agenda of the next IAEA SCNL meeting, thus putting a (provisional) end to the exercise.

Although the Vienna Convention eventually entered into force on 12 November 1977, the IAEA SCNL, at its fifth session, in 1978, confined itself to a brief discussion of the ways to promote wider acceptance thereof, in particular by States having significant nuclear activities. As to the relationship between the Vienna Convention and Paris Convention, the IAEA SCNL merely expressed a desire that "every effort should be made to achieve increased harmonization" between the two Conventions; it was also pointed out that "means should be explored to facilitate the simultaneous application of both".³⁴ No further attention appears to have been paid to the 1974 Draft Joint Protocol until, in May 1984, the IAEA SCNL resumed examination of the relationship between the Vienna Convention and the Paris Convention, and again raised the question of the desirability of establishing a formal link between the two Conventions.³⁵ Subsequently, at the NEA Group meeting of November 1985, the IAEA representatives stated that the

²⁸ OECD-NEA document SEN(75)1.

²⁹ OECD-NEA document SEN(75)1, para. 16.

³⁰ OECD-NEA document SEN(75)1, para. 17.

³¹ OECD-NEA document SEN(75)12, p. 14.

³² All of these questions are addressed in Section 2, during the examination of the legal nature of the Joint Protocol and the implications of Articles III and IV thereof.

³³ OECD-NEA document SEN(75)12.

³⁴ IAEA document CN-12/SC-4/5, para. 9.

³⁵ IAEA document NS-TC-462.5, paras 14–15.

time had come to reactivate consideration of the matter, as additional States were considering ratification of, or accession to, the Vienna Convention and North–South bilateral nuclear cooperation and supply arrangements were increasing. Consequently, the NEA Group endorsed the IAEA representatives' proposal for a joint study of the relationship between the two Conventions.³⁶

A joint Note was consequently produced by the NEA and IAEA Secretariats in May 1986 summarizing the work done in the 1970s and recommending the reassessment of the desirability of the solution envisaged at the time — that is, the adoption of a joint protocol aimed at "harmonizing the coexistence of the two Conventions". This Note was thereafter presented as a working paper for a Joint IAEA/NEA Informal Meeting of Experts which met in Vienna on 8–10 September 1986.³⁷ Attached to the Note were the text of the 1974 Draft Joint Protocol,³⁸ together with a new Explanatory Note (hereinafter referred to as the "1986 Explanatory Note")³⁹ and an Analytical Table of the Articles of the Vienna Convention and the Paris Convention in the Context of the Draft Joint Protocol⁴⁰. At the September 1986 meeting, the experts reviewed the five options that had been envisaged in WD II, plus a sixth option consisting in "extension of the territorial scope of both Conventions". However, they concluded that the option of a joint protocol establishing a formal relationship between the two Conventions was "the most practicable and effective solution", since it would both: (a) "extend the benefit of the special regime of liability for nuclear damage set up under each Convention in order to provide for wider protection of victims", and (b) "eliminate problems arising from the simultaneous application of both Conventions to a nuclear incident". After reviewing the 1974 Draft Joint Protocol, the experts suggested a number of improvements thereto. Finally, the experts also suggested that discussions concerning a possible revision of the Vienna Convention would provide an opportunity to concurrently examine the question of a joint protocol.⁴¹

In this latter respect, the Chernobyl accident in April 1986 had in fact again raised the issues of ensuring a more widely accepted nuclear liability regime and of making that regime more adequate to cope with the transboundary consequences of a major nuclear incident. The discussions initiated within the IAEA in order to cope with these issues eventually led, on the one hand, to the adoption of the 1997 Vienna Protocol and, on the other, to the adoption of an entirely new Convention, the 1997 Convention on Supplementary Compensation for Nuclear Damage (CSC), designed to create a truly worldwide nuclear liability regime and to increase the amount of compensation available in the event of nuclear damage. The CSC, in particular, is designed to create a treaty link between not only the Contracting Parties to either the Paris Convention or the Vienna Convention (including any amendments thereto), but also other States party to neither Convention; it therefore has an even wider scope than the 1988 Joint Protocol.⁴² However, in the early stages of the discussions that led to the adoption of these later Conventions, the decision was taken to first revive the earlier efforts to create a treaty link between the Contracting Parties to the Paris Convention.

The suggestions made by the September 1986 Joint IAEA/NEA Informal Meeting were thus brought to the attention of, on the one hand, the NEA Group and, on the other, the IAEA SCNL. In November 1986, the NEA Group endorsed the suggestions made in September.⁴³ In March 1987, an open ended session of the IAEA SCNL was convened and the IAEA SCNL, in its turn, discussed and endorsed the solution of a joint protocol. The IAEA SCNL was presented with the 1974 Draft Joint Protocol and an Explanatory Note almost identical to the 1986 Explanatory Note⁴⁴, approved changes to the 1974 Draft Joint Protocol consisting in the elaboration of a draft preamble and the amendment of its operative provisions, and, finally, recommended that completion of work be aimed at, either by reconvening the IAEA SCNL in the future or by establishing a Joint IAEA/NEA Working Group.⁴⁵ This latter solution

³⁶ OECD-NEA document SEN/LEG(86)1, paras 65-66.

³⁷ IAEA document 250-N5.52.14.

³⁸ IAEA document 250-N5.52.14, Appendix I.

³⁹ IAEA document 250-N5.52.14, Appendix II. The 1986 Explanatory Note was later reproduced almost verbatim in a corresponding Explanatory Note presented to the IAEA SCNL in March 1987 (IAEA document N5.TC-462.6/3).

 ⁴⁰ IAEA document 250-N5.52.14, Appendix III. The Analytical Table was also later reproduced in IAEA document N5.TC-462.6/3.
 ⁴¹ IAEA document N5.TC-462.6/2.

⁴² IAEA International Law Series No. 3. Section 3 is devoted, in particular, to the CSC.

⁴³ OECD-NEA document SEN/LEG(87)1, paras 18 and 36.

⁴⁴ IAEA document N5.TC-462.6/3.

⁴⁵ IAEA document 250-N.5.TC-462.6.

was agreed to in principle by the NEA Steering Committee in April 1987⁴⁶ and by the IAEA Board of Governors in June 1987⁴⁷.

Before the Joint IAEA/NEA Working Group met, the NEA Group met again in June 1987 in order to examine the outcome of the IAEA SCNL, and examined a Note on the "Draft Joint Protocol Relating to the Application of the Paris Convention and the Vienna Convention" prepared by the Secretariat in order to synthesize the essential elements of the working documents distributed in the past that had led to the preparation of the 1974 Draft Joint Protocol and to reflect the evolution in thinking since then.⁴⁸ The NEA Group agreed that, subject to appropriate adjustments, that Note could be used as a basis for a joint document of the NEA and IAEA Secretariats to be presented to the Joint IAEA/NEA Working Group meeting; in addition, it produced an alternative text for the operative provisions of the Joint Protocol.⁴⁹ Consequently, when the Joint IAEA/NEA Working Group of Governmental Experts on the Relationship between the Paris and the Vienna Convention met in Vienna, on 27–30 October 1987, it was presented with the draft preamble and articles adopted in principle by the IAEA SCNL in March 1987, the drafting proposals discussed by the NEA Group in June 1987, and a document prepared by the IAEA and NEA Secretariats of the IAEA and the NEA were also presented to the meeting for discussion.⁵¹

The Joint IAEA/NEA Working Group adopted by consensus a Final Document with a new draft text (hereinafter referred to as the "1987 Draft Joint Protocol"), which later became the definitive text of the Joint Protocol.⁵² This text was later approved by the NEA Group at its meeting in November 1987. At the same time, the NEA Group recommended to the Steering Committee that, subject to approval by the IAEA Board of Governors, the required preparations be made for the organization of a 'Diplomatic Conference' in conjunction with the IAEA General Conference in 1988 to adopt the Joint Protocol as agreed upon in the Joint Meeting. Finally, the NEA Group requested the two Secretariats to prepare an explanatory note for the governing bodies of the two organizations, in view of the fact that "the rather simple text of the Protocol might easily lead to erroneous conclusions".⁵³ This Explanatory Note (hereinafter referred to as the "1987 Explanatory Note") was produced in December 1987.⁵⁴

The 1987 Draft Joint Protocol, together with the 1987 Explanatory Note, was examined by the IAEA Board of Governors in February 1988. The Board endorsed the 1987 Draft Joint Protocol and agreed to convene a one-day conference to be organized jointly by the IAEA and the NEA in conjunction with the 32nd regular session of the IAEA General Conference for the purpose of adopting the Joint Protocol and opening it for signature.⁵⁵ A similar decision was later taken, upon the recommendation of the NEA Steering Committee, by the OECD Council in June 1988.⁵⁶ Consequently, a diplomatic conference was jointly convened by the two organizations in Vienna on 21 September 1988.⁵⁷ The Diplomatic Conference adopted the text of the Joint Protocol by consensus. Nineteen States signed the

⁴⁶ OECD-NEA document NE/M(87)1, para. 9(b).

⁴⁷ IAEA document GOV/2305, and IAEA document GOV/OR.675.

⁴⁸ OECD-NEA document LEG/DOC(87)3 (Rev. 1).

⁴⁹ OECD-NEA document SEN/LEG(87)3.

⁵⁰ IAEA document N5/TC/643 (4). This was in fact the envisaged revision of OECD-NEA document LEG/DOC(87)3.

⁵¹ IAEA document N5/TC/643 (4), Room Document No. 1 (attached).

⁵² IAEA document N5/TC/643, Final Document (attached). The Draft Joint Protocol was later reproduced in Annex I to IAEA document GOV/2326.

⁵³ OECD-NEA document SEN/LEG(88)1.

⁵⁴ The 1987 Explanatory Note was then reproduced in Annex II to IAEA document GOV/2326.

⁵⁵ IAEA document GOV/2326, and IAEA document GOV/OR.686.

⁵⁶ OECD document C/M(88)12.

⁵⁷ Conference on the Relationship between the Paris and Vienna Conventions, Official Record of a Meeting held at the Vienna International Centre on Wednesday, 21 September 1988, at 10.15 a.m., IAEA, Vienna (1988).

Joint Protocol on the very day of its adoption, but the Joint Protocol remained open for signature until the date of its entry into force on 27 April 1992 (i.e. three months after the date of deposit of instruments of ratification or accession by at least five States party to the Paris Convention and five States party to the Vienna Convention). In 1989, a short publication was issued on behalf of both the IAEA and the NEA containing the Final Act of the Diplomatic Conference, together with the text of the Joint Protocol and the 1987 Explanatory Note.⁵⁸

⁵⁸ Nuclear Liability: Joint Protocol Relating to the Application of the Vienna Convention and the Paris Convention, 1988, IAEA, Vienna (1989).

2. GENERAL ASPECTS

2.1. THE LEGAL NATURE OF THE JOINT PROTOCOL AND ITS RELATION TO THE VIENNA CONVENTION AND THE PARIS CONVENTION

As is pointed out in Section 1.3, one of the options envisaged in the 1970s for creating a treaty link between the Contracting Parties to either the Vienna Convention or the Paris Convention was to adopt "an instrument or instruments whereby a 'bridge' is created between the two Conventions and whereby all Parties...to the Vienna Convention would be treated...as if they were Parties to the Paris Convention...and vice versa"⁵⁹. This solution was eventually preferred to the others for various reasons, among which was that it would "leave the two Conventions intact so that there will be no reopening of certain issues regarding the substance of the two Conventions" and, at the same time, the envisaged instrument(s) "could be limited to relatively few Articles and the changes required in the legislation of the Parties would be limited."⁶⁰

Although the 'bridge' between the two Conventions could have consisted of either a single instrument (additional to both Conventions) or two separate instruments (each additional to one Convention), the former solution was eventually preferred, again for various reasons. In the first place, it was thought that it "would stress the element of reciprocity", since the Parties to one Convention would undertake to treat the Parties to the other Convention as if they were Parties to the former Convention and, at the same time and "as consideration therefor", the Parties to the other Convention would undertake to afford the same treatment to the Parties to the former Convention. In addition, the adoption of a single joint instrument would entail certain practical advantages, such as the need for only one diplomatic conference. As for its designation, although it was considered that the term 'protocol' (or 'additional protocol') would have better suited the adoption of two distinct instruments, whereas the term 'convention' would have been more appropriate for a single instrument, it was also recognized that the question was not one of substance and the term 'protocol' was eventually preferred.⁶¹

The Joint Protocol is, therefore, an international agreement or treaty which is, on the one hand, distinct from the Vienna Convention and the Paris Convention and, on the other, additional to both. Inasmuch as it is intended to create a 'bridge' between the two Conventions, whereby the Parties to either Convention are to be treated as if they were Parties to the other, the Joint Protocol is not a 'free-standing' international agreement and, under Articles VI and IX, is only open to States which are already Parties to either the Vienna Convention or the Paris Convention and only so long as they remain Parties thereto.⁶²

Both the Vienna Convention and the Paris Convention are defined in similar terms in Article I of the Joint Protocol in order to include not only the original version, but also "any amendment thereto which is in force for a Contracting Party" to the Joint Protocol: this definition, which did not appear in the 1974 Draft Joint Protocol, was first provisionally adopted by the Joint IAEA/NEA Working Group in October 1987 and thereafter remained unchanged until its final adoption at the Diplomatic Conference in 1988. As is recalled in Section 1.1, the Paris Convention was amended by two Additional Protocols adopted respectively in 1964 and 1982, and will be further amended by a Protocol adopted in 2004, which, however, is not yet in force, whereas the Vienna Convention has been amended only once as a result of the entry into force of the 1997 Vienna Protocol.

⁵⁹ WD I, para. III.1, and WD II, para. 46.

⁶⁰ WD I, para. III.1, and WD II, para. 47. It was, however, recognized that this solution would add another instrument (or instruments) to the existing Conventions and "might create complications by adding a 'third class' of countries, i.e. those that ratify the instrument(s), to the existing two, i.e. the Parties of the Paris and of the Vienna Convention", and it was pointed out that this would have to be borne in mind when examining the provisions concerning the entry into force of the instrument(s) (WD I, para. III.1, and WD II, para. 48). This latter issue is touched upon in Section 4.3.

⁶¹ WD I, para. III.2, and WD II, paras 49–50.

⁶² On this issue, see also Section 4.2.

The adoption of the 1997 Vienna Protocol has potentially complicated the legal framework in which the Joint Protocol is called to operate, since, unlike the successive Protocols purporting to amend the Paris Convention, the 1997 Vienna Protocol was conceived as a distinct treaty, to some extent independent of the original Vienna Convention.⁶³ There can be little doubt that a Party to the Vienna Protocol, which is expressly intended to "amend" the 1963 Vienna Convention, is entitled to become a Party to the Joint Protocol under Article I thereof. However, the Parties to the Vienna Protocol need not necessarily be in treaty relations with the Parties to the original 1963 Vienna Convention:⁶⁴ consequently, although in practice this has not happened so far, the Joint Protocol is now potentially called upon to create a bridge between three, rather than two, distinct treaty regimes.⁶⁵

2.1.1. The question of whether the Joint Protocol constitutes an amendment of the Vienna Convention and the Paris Convention

The Joint Protocol is not an amending treaty in respect of either the Vienna Convention or the Paris Convention, whose provisions remain thereby formally unaffected. Although, in the early stages of its elaboration, some concerns were raised as to whether the projected new treaty would amount to an amendment of the two Conventions, the view eventually prevailed that neither its procedural provisions nor its operative rules would legally amount to such an amendment.

As far as the procedural provisions are concerned, WD III points out, in respect of the 1974 Draft Joint Protocol, that: "If that were the case, a separate Protocol to each Convention would in fact be necessary." WD III goes on to say that:

"it would be necessary to amend the Conventions if it were required that all future ratifications of, and accessions to, either Convention be accompanied by ratification of, or accession to, the Protocol. The Draft Joint Protocol does not contain this requirement [nor, indeed, does the Joint Protocol as adopted in 1988]. Instead, Article V [Article VII of the Joint Protocol] stipulates that a sufficiently high number of Parties to either Convention must have deposited their instruments of ratification before the Protocol will enter into force; Articles IV(b) and VII [Articles VI.1 and IX of the Joint Protocol] ensure that only Parties to either Convention may become and remain Parties to the Protocol. These procedural Articles of the Protocol change neither the text nor the substance of either Convention."⁶⁶

⁶³ Amendments to the Paris Convention have been adopted in accordance with Article 20 thereof, whereby the amendments, once adopted, come into force when ratified or confirmed by two thirds of the Contracting Parties, and, for each Contracting Party ratifying or confirming later, at the date of such ratification or confirmation. Therefore, only the Contracting Parties to the Paris Convention are entitled to ratify or confirm the amendments so adopted. In contrast, under its Article 20, the Vienna Protocol is open to all States, not just the Contracting Parties to the 1963 Vienna Convention.

⁶⁴ Under Article 19 of the Vienna Protocol, a State which is a Party to the 1997 Vienna Protocol but not to the 1963 Vienna Convention shall be bound by the Vienna Convention as amended by the Vienna Protocol in relation to other States Parties to the Vienna Protocol. In addition, a State which is a Party to the 1997 Vienna Protocol but not to the 1963 Vienna Convention shall also be bound by the 1963 Vienna Convention in relation to the States which are only Parties thereto, but only "failing an expression of a different intention" at the time when it signs, ratifies or accedes to the Vienna Protocol.

⁶⁵ So far, no Contracting Party to the 1997 Vienna Protocol has expressed the intention not to be in treaty relations with the Contracting Parties to the 1963 Vienna Convention. In a (theoretical) situation where there were no treaty relations between a Contracting Party to the 1963 Vienna Convention and a Contracting Party to the 1997 Vienna Protocol, such treaty relations could not be established by virtue of the Joint Protocol alone in the case that both States were Parties thereto: the Joint Protocol is intended to establish treaty relations between Parties to the Vienna Convention and Parties to the Paris Convention, and not between Parties to different versions of the Vienna Convention.

⁶⁶ OECD-NEA document SEN(75)1, para. 18. WD III also points to para. 6 of the 1974 Explanatory Note.

As for the operative provisions of the projected new treaty, WD III further points out that:

"The same is true for its operative Articles I and II [Articles IV and III of the Joint Protocol]. The Parties to the Protocol and either Convention accord to the Parties of the other Convention and the Protocol the status of Contracting Parties by means of a separate international agreement which is completely outside both Conventions. The Convention determined as applicable under the rule of Article II of the Draft Joint Protocol [Article III of the Joint Protocol] will be applied without any change. The Parties to either Convention which do not ratify the Protocol will not suffer any disadvantage or discrimination vis à vis the Parties to the Protocol which neither affects the enjoyment of their rights nor the performance of their obligations under the Conventions. The interests of all States entitled to become Parties to either Convention could be safeguarded by inviting them to the diplomatic conference for the adoption of the Joint Protocol."⁶⁷

During the discussion of WD III that took place within the NEA Group in March 1975, France specifically raised the question of whether the extension of the operator's liability under the Vienna Convention to cover damage suffered in the territory of Contracting Parties to the Paris Convention — which was then envisaged in Article I of the 1974 Draft Joint Protocol, corresponding to Article IV of the Joint Protocol, but which is now more clearly spelled out in Article II of the Joint Protocol — would amount to an amendment to Article VIII.3 of the Vienna Convention whereby, according to France's interpretation, "the funds provided by insurance or otherwise should be exclusively available for compensation due under the Vienna Convention", and the Netherlands expressed the view that this was in fact the case. However, the NEA Secretariat explained that:

"States Party to the Vienna Convention were free, at least according to some views, to extend by national legislation the operator's liability to incidents occurring and nuclear damage suffered in non-Contracting States. If that were the case, the proposed Protocol would not affect Article VII.3 of the Vienna Convention."⁶⁸

On the same issue, the background materials that were at the basis of the definitive adoption of Articles IV and II of the Joint Protocol confirmed the view thus expressed and in respect of both the Vienna Convention and the Paris Convention.⁶⁹

2.2. THE OBJECT AND PURPOSE OF THE JOINT PROTOCOL AS EXPRESSED IN ITS PREAMBLE

As is often the case with international treaty instruments, the Preamble of the Joint Protocol briefly explains the reasons for the adoption of the treaty, as well as its object and purpose. The original 1974 Draft Joint Protocol did not have a preamble. The current Preamble is the result of a draft which was first examined and endorsed by the IAEA SCNL at its sixth session, in March 1987; the Preamble was then provisionally adopted by the Joint IAEA/NEA Working Group in October 1987 and remained unchanged until its final adoption at the Diplomatic Conference in 1988.⁷⁰

⁶⁷ OECD-NEA document SEN(75)1, para. 18.

⁶⁸ OECD-NEA document SEN(75)12, pp. 11–12.

⁶⁹ See, in particular, OECD-NEA document LEG/DOC(87)3 (Rev. 1), para. 46, where it is pointed out that, although "the Protocol could lead to enlarging the number of victims entitled to compensation, to the detriment of victims suffering damage in the territories of Contracting Parties to either Convention which are not Parties to the Protocol", this consequence "does not seem to modify the text of the Conventions. The [Paris Convention] (and also the [Vienna Convention], according to some authors) may be extended by Contracting Parties to nuclear damage suffered in non-Contracting Parties without the other Contracting Parties having to consent thereto and the Joint Protocol does not change this situaton". It was also pointed out that "the Protocol establishes the principle of channelling between [Contracting Parties to the Paris Convention] and [Contracting Parties to the Vienna Convention]". See also IAEA document N5/TC/643(4), para. 45.

⁷⁰ IAEA document 250-N.5.TC-462.6, and IAEA document N5/TC/643.

In the Preamble, the Contracting Parties to the Joint Protocol declare themselves "desirous to establish a link" between the Paris Convention and the Vienna Convention on the basis, on the one hand, of the consideration that no State was a Party to both Conventions at the time of its adoption (a fact that has since remained unchanged) and, on the other, of their conviction that "adherence to either Convention by Parties to the other Convention could lead to difficulties resulting from the simultaneous application of both Conventions to a nuclear incident".

Having thus briefly explained the reason why the Joint Protocol was adopted, the Contracting Parties further clarify that the principal purpose of the establishment of a treaty link between the Vienna Convention and the Paris Convention is the mutual extension of "the benefit of the special regime of civil liability for nuclear damage set forth under each Convention". At the same time, they also make it clear that the Joint Protocol is intended to "eliminate conflicts arising from the simultaneous application of both Conventions to a nuclear incident". As was explained in the 1987 Explanatory Note, it can, therefore, be said that the Joint Protocol has a "dual purpose" and is based on two underlying principles.⁷¹

The first principle, which underlines Articles II and IV of the Joint Protocol, is "to create a link or 'bridge' between the two Conventions by abolishing the distinction between Contracting Parties and non-Contracting States as regards the operative provisions of either Convention."⁷² Of course, this does not mean that, by becoming Parties to the Joint Protocol, the Contracting Parties to either Convention become Contracting Parties to the other Convention. Rather, it means that the Contracting Parties to either Convention and the Joint Protocol undertake to apply the operative provisions of that Convention with respect to the Contracting Parties to both the other Convention and the Joint Protocol in the same manner as those provisions apply between Contracting Parties to the same Convention.

The second principle, which underlines Article III of the Joint Protocol, consists in making either the Vienna Convention or the Paris Convention exclusively apply to a nuclear incident by means of an "appropriate conflict rule."⁷³

⁷¹ 1987 Explanatory Note, para. 5.

⁷² 1987 Explanatory Note, para. 4.

⁷³ 1987 Explanatory Note, para. 4.

3. OPERATIVE PROVISIONS

3.1. THE MUTUAL EXTENSION OF THE OPERATOR'S LIABILITY UNDER EITHER CONVENTION TO DAMAGE SUFFERED IN THE TERRITORY OF CONTRACTING PARTIES TO THE OTHER CONVENTION (ARTICLE II)

As is pointed out in Section 1.2, the first problem originating from the absence of treaty relations between the Contracting Parties to different Conventions relates to the so-called geographical scope of the nuclear liability regime, inasmuch as the Convention under which the operator is liable may not apply to nuclear damage suffered in the territory of the Contracting Parties to a different Convention.

The Paris Convention, as at present in force, applies neither to nuclear incidents occurring in non-Contracting States nor to nuclear damage suffered in such States, unless otherwise provided by the legislation of the Installation State (Article 2). Although, on 22 April 1971, the OECD Steering Committee recommended to the Contracting Parties that they extend the application of the Paris Convention to damage suffered in Contracting States (or on the high seas on board a ship registered in the territory thereof) even if the nuclear incident occurs in a non-Contracting State, it did not recommend a similar extension to damage suffered in non-Contracting States.⁷⁴ Once the 2004 amending Protocol enters into force, under the amended Article 2, the Paris Convention will apply to damage in non-nuclear non-Contracting States, but will only apply to damage suffered in nuclear non-Contracting States if these States have legislation in place conforming to the basic principles of nuclear liability of the Paris Convention and affording "equivalent reciprocal benefits".

Similarly, although the 1963 Vienna Convention contains no express provision as to its territorial scope, at its very first session, in April 1964, the IAEA SCNL took the view that the Convention is applicable even if the nuclear incident occurs on the high seas or in the territory of a non-Contracting State, but only in respect of damage suffered within the jurisdiction of the Contracting Parties or on the high seas.⁷⁵ The 1997 Vienna Convention applies in principle to nuclear damage "wherever suffered", but then allows the legislation of the Installation State to exclude damage suffered in a nuclear non-Contracting State not affording "equivalent reciprocal benefits" (Article I A of the 1997 Vienna Convention).

The Joint Protocol is intended to make sure that, in relations between the Contracting Parties to the Protocol, there is a mutual extension of the operator's liability under one Convention to damage suffered in territories of Parties to the other Convention. In the 1974 Draft Joint Protocol there was no express provision to this effect, but the same principle was implicit in the more general provision whereby the Parties to either the Paris Convention or the Vienna Convention were to be "considered as if they were Parties" to the other Convention in respect of the application of the operative provisions thereof, including those relating to the scope of the operator's liability (Article I of the 1974 Draft Joint Protocol, corresponding to what is now Article IV of the Joint Protocol, which is discussed in the next section). However, at the June 1987 NEA Group meeting, the NEA Secretariat explained the reasons that had led it "to replace the abstract provisions of the 1974 Draft Joint Protocol with more concrete ones" and, in particular, to propose a new provision which would clearly state "the main substantive rule of the Protocol, mainly the territorial extension of both Conventions", and the experts agreed with that approach.⁷⁶ As was rightly pointed out in the NEA Secretariat Note of

⁷⁴ Paris Convention: Decisions, Recommendations, Interpretations, OECD Nuclear Energy Agency, Paris (1990), pp. 9–10.

⁷⁵ IAEA document CN-12/SC/9, paras 4–5. In the course of the discussions that eventually led to the adoption of the Joint Protocol, the question arose within the NEA Group as to the legal force of the view expressed by the IAEA SCNL in 1964. In particular, when the question was raised by Germany at the March 1975 meeting, the representative of the IAEA Secretariat quoted from the record of the IAEA SCNL's first series of meetings (IAEA document CN-12/SC/9, para. 3) that "It was the understanding of the Committee that the conclusions it reached, as set forth in Chapter IV, could not be regarded as an interpretation of the Convention which would be binding upon the parties thereto". However, the IAEA representative added that, "Although not legally binding, this interpretation would presumably have a strong authoritative force" (OECD-NEA document SEN(75)12, para. V).

⁷⁶ OECD-NEA document SEN/LEG(87)3, paras 32–33.

May 1987, as well as in the "Background Material" thereafter presented to the October 1987 Joint IAEA/NEA Working Group meeting, this new approach was the result of a "change of emphasis" as far as the objectives of the Joint Protocol were concerned: in the early 1970s, when the work on the relationship between the Vienna Convention and the Paris Convention began, the principal objective pursued was the "avoidance of conflicts between the two Conventions from which operators, insurers and carriers would benefit in the first place", as well as the "wider acceptance of the basic system underlying both Conventions". The aim of providing compensation for victims of a nuclear incident occurring in the territories of Contracting Parties to either Convention "was emphasized as well, but it did not become the essential preoccupation until after the Chernobyl accident", an event which "triggered public concern about international civil liability regimes for nuclear damage and…made lawmakers aware of the need not only to enlarge this system but also to state the objectives of such enlargement as clearly as possible".⁷⁷

Having thus decided to place the emphasis on directly spelling out the extension of the liability and compensation system of both Conventions, the NEA Group, at the June 1987 meeting, discussed two alternative ways to provide therefor: one was to provide for the extension of the scope of application of either Convention to cover nuclear damage suffered in the territory of one or more Contracting Parties to the other Convention (alternative 1); the other was to stipulate that nuclear operators would be liable for such damage according to the Convention to which their Installation State is a Party (alternative 2)⁷⁸. Although both alternatives would have achieved the result of extending the benefits afforded by either Convention, the NEA Secretariat pointed out that alternative 1, by focusing on the territorial scope of application of either Conventions, the NEA Secretariat pointed out that alternative 1, by focusing on the territorial scope of application of either Conventions, the NEA Secretariat pointed out that alternative 1, by focusing on the territorial scope of application of either Conventions, the NEA Secretariat pointed out that alternative 1, by focusing on the territorial scope of application of either Conventions, the NEA Secretariat pointed out that alternative 1, by focusing on the territorial scope of application of either Conventions, the NEA Secretariat pointed out that alternative 1, by focusing on the territorial scope of application of either Conventions, the NEA Secretariat pointed out that alternative 1, by focusing on the territorial scope of application of either Conventions, the NEA Secretariat pointed out that alternative 1, by focusing on the territorial scope of application of either Conventions, the NEA Secretariat pointed out that alternative 2, which alternative was preferable, and both were provisionally retained for consideration by the Joint IAEA/NEA Working Group.⁸⁰ At its meeting in October 1987, the Joint IAEA/NEA Working Group eventually decided to opt for alternative 2,

Thus Article II of the Joint Protocol explicitly states that, for the purpose of the Joint Protocol, the operator of a nuclear installation situated in the territory of a Party to either the Paris Convention or the Vienna Convention shall be liable in accordance with that Convention for nuclear damage suffered in the territory of a Party to the other Convention and the Joint Protocol. This mutual extension of the territorial scope of the operator's liability does not, however, otherwise affect that liability as determined by the applicable Convention: the operator remains liable under the Convention to which the State in which his installation is situated is a Party, and the amount of his liability is still determined by the legislation of that State pursuant to that Convention.⁸²

Interestingly, Article II of the Joint Protocol says nothing of the place of the nuclear incident. The original draft Article I proposed by the NEA Secretariat in 1987 contained the proviso that the nuclear incident causing the damage must have occurred in the territory of a Contracting Party to either the Vienna Convention or the Paris Convention that is also a Party to the Protocol, but only in respect of alternative 2,⁸³ and the NEA Group decided to insert it in alternative 1 as well.⁸⁴ The experts apparently felt that such a proviso was needed in order to "specify that the Joint Protocol as such does not cover nuclear damage suffered in the territories of its Contracting Parties by incidents occurring in non-Contracting States, i.e. in those States which are not Party to either Convention nor to the Protocol", this being "a matter for national legislation outside the Protocol", although they did recall that the OECD Steering Committee had adopted a recommendation to extend the Paris Convention to such damage. On the other hand, they pointed out that, in the June 1987 draft, no mention was made of nuclear incidents occurring and nuclear damage

⁷⁷ OECD-NEA document LEG/DOC(87)3 (Rev. 1), para. 9, and IAEA document N5/TC/643(4), para. 9.

⁷⁸ OECD-NEA document LEG/DOC(87)3 (Rev. 1), Annex IV.

⁷⁹ OECD-NEA document SEN/LEG(87)3, para. 32.

⁸⁰ OECD-NEA document SEN/LEG(87)3, para. 34.

⁸¹ IAEA document 250-N5-TC-643.

⁸² See Sections 3.2.3 and 3.3.

⁸³ OECD-NEA document LEG/DOC(87)3 (Rev. 1), Annex IV.

⁸⁴ OECD-NEA document SEN/LEG(87)3, para. 34.

suffered on or over the high seas, "as there seems to be general agreement that both Conventions apply in such cases".⁸⁵ This observation probably explains why all reference to the place of the nuclear incident was eventually dropped when the definitive text of Article II was adopted, since the June 1987 proviso would have excluded, inter alia, incidents on the high seas causing damage in Contracting Parties to the Joint Protocol.

3.2. THE ELIMINATION OF THE DISTINCTION BETWEEN CONTRACTING PARTIES AND NON-CONTRACTING STATES AS REGARDS THE APPLICATION OF THE OPERATIVE PROVISIONS OF EITHER CONVENTION (ARTICLE IV)

The 1987 Explanatory Note described Article IV as a "complementary" provision with respect to the other operative provisions in the Joint Protocol, namely, Article II, which is examined in Section 3.1 and relates to the mutual extension of the operator's liability to damage suffered in the territory of Contracting Parties to the Joint Protocol, and Article III, which is examined in Section 3.3 and relates to the choice of the applicable Convention. This may well be an accurate description of the function of Article IV in the context of the Joint Protocol as it emerged from the discussions held in 1987. However, it must not be overlooked that, as the 1987 Explanatory Note itself points out, the first purpose of the Joint Protocol still is to "create a link or 'bridge' between the two Conventions by abolishing the distinction between Contracting Parties and non-Contracting States as regards the operative provisions of either Convention³⁸⁶. Indeed, as is pointed out in Section 3.1, in the original 1974 Draft Joint Protocol even the mutual extension of the operator's liability under one Convention to damage suffered in territories of Parties to the other Convention, which is now expressly provided for in Article II of the Joint Protocol, was implicit in the more general provision whereby the Parties to either the Paris Convention or the Vienna Convention were to be "considered as if they were Parties" to the other Convention in respect of the application of the operative provisions thereof. That provision, which, significantly, opened the 1974 Draft Joint Protocol as Article I, was the basis of Article IV of the Joint Protocol. Therefore, despite the change of emphasis that occurred as a result of the Chernobyl accident, which resulted in the direct spelling out of the extension of the liability and compensation system of both Conventions in Article II of the Joint Protocol, it seems fair to say that Article IV can still be regarded as a fundamental provision, if not the core provision, of the Joint Protocol.

As is pointed out in Section 1.2, the problems which the Joint Protocol was intended to solve derive from the very existence of distinct treaty systems which, although largely similar in content, have different Contracting Parties. When the idea of a treaty instrument creating a link between the Vienna Convention and the Paris Convention was first envisaged in the early 1970s, a thorough discussion was undertaken in the working documents in order to determine under which conditions and to what extent the Parties to one Convention are to be treated as if they were Parties to the other Convention. This discussion explains the choices made in Article I of the original 1974 Draft Joint Protocol, the substance of which largely corresponds to Article IV of the Joint Protocol.

First, regarding the question as to which States should be treated as if they were Parties to the Conventions, WD I and WD II point out that a solution whereby all (present and future) Parties to either Convention would be treated as if they were Parties to the other, regardless of whether they were Parties to the Protocol, "could, at least partially, result in a unilateral treatment as Party". In this respect, it was further pointed out that, according to the principle of international law that a treaty does not create either obligations or rights for a third party without its consent,⁸⁷ such a unilateral treatment would not be automatically binding on the non-Parties to the Protocol, which would have to give their consent

⁸⁵ IAEA document N5/TC/643(3), p. 4. This document, which was presented to the October 1987 Joint IAEA/NEA Working Group meeting, contains the "Drafting proposals discussed by the NEA Group ... in June 1987", together with "Comments" relating thereto which were discussed by the NEA Group at the same meeting and which were included in the IAEA document at the request of the NEA Secretariat.

⁸⁶ 1987 Explanatory Note, para. 4.

⁸⁷ 1969 Vienna Convention on the Law of Treaties, Article 34.

or assent.⁸⁸ Moreover, the absence of a condition of reciprocity might provide little incentive for ratification of the Protocol. Therefore, the alternative solution, whereby only the Parties to the Protocol would be treated as if they were Parties to the Convention in question, was eventually preferred as being "truly reciprocal". This solution was, therefore, reflected in Article I of the original 1974 Draft Joint Protocol and is still reflected in Article IV of the Joint Protocol.⁸⁹

As to the substance of the problem, WD I and WD II point out that two basic solutions might be envisaged. The first consisted in merely aiming at "resolving problems caused by the simultaneous application of Article II.1 (b) (iv) and (c) (iv) of the Vienna Convention and Article 4 (a) (iv) and (b) (iv) of the Paris Convention and would limit the treatment as parties to those provisions". Thus, the mutual extension of the operator's liability under one Convention to damage suffered in the territories of Parties to the other Convention was clearly not considered to be the main issue at that time; rather, the main issue was considered to be the determination of the operator liable in transport cases, since, as is pointed out in Section 1.2, the identical rules laid down in both Conventions differentiate between transport taking place between persons situated within the territories of Contracting Parties and transport taking place between one person situated in the territory of a Contracting Party and another person situated in the territory of a non-Contracting State. In any event, WD I and WD II further point out that "Such a limited solution might…not be adequate, since a number of problems might still remain". It was mentioned, inter alia, that, if that limited solution were adopted, "It could be argued that the jurisdictional Articles would not be covered" by the Protocol. As a result, the second solution, providing, in principle, that "the Parties to either Convention would be generally treated as if they were Parties to both Conventions", though only in respect of the operative provisions thereof, was eventually preferred.⁹⁰

In Article I of the original 1974 Draft Joint Protocol, this question was resolved by enumerating the inapplicable articles of each Convention, namely, Articles XVI, XVII and XXI–XXVI of the Vienna Convention and Articles 6(e), 7(e) and 17–22 of the Paris Convention. Interestingly, the enumeration of the inapplicable articles did not exclusively refer to the "procedural" provisions in either Convention relating to such matters as signature, ratification, accession, etc., as had been envisaged in WD I and WD II. The 1974 and 1986 Explanatory Notes, both concerning the 1974 Draft Joint Protocol, clarified the choice made in that draft. Since both Explanatory Notes are identical in substance, only the 1986 Explanatory Note will be quoted here:

"From an Article-by-Article analysis of both Conventions, it would appear that three categories of provisions can be distinguished for the purposes of the Joint Protocol:

- (i) Provisions which should be made applicable, i.e. those which refer to the concept of non-Contracting States, proposed to be eliminated under the Joint Protocol;
- (ii) Provisions which are not directly relevant to the concept of non-Contracting States and which are identical in substance to those of the other Convention; these provisions would be applicable pursuant to Article II of the Joint Protocol [corresponding to Article III of the Joint Protocol relating to the applicable Convention], in particular, if referred to in the applicable provisions of either Convention, or if they set out general concepts or definitions ...[;]
- (iii) Provisions which are not applicable: these include firstly, the procedural provisions of either Convention dealing with signature, ratification, accession, entry into force, etc.; secondly, those provisions which are either peculiar to each Convention (e.g. Articles 7(e) and 17 of the Paris Convention and Article XVI of the Vienna Convention) or are different in substance (Article 6(e) of the Paris Convention)".⁹¹

⁸⁸ 1969 Vienna Convention on the Law of Treaties, Articles 35 and 36.

⁸⁹ WD I, para. III. 3, and WD II, paras. 57-63.

⁹⁰ WD I, para. III. 3, and WD II, paras. 57-63.

⁹¹ 1986 Explanatory Note, para. 3.

At its March 1987 meeting, however, the IAEA SCNL decided to change the approach taken in the 1974 Draft Joint Protocol and proposed a new text based on the opposite approach: Article I of the 1974 Draft Joint Protocol became Article II, and the enumeration of the inapplicable Articles was replaced by a positive formula whereby the applicable articles of either Convention, that is, Articles I–XV of the Vienna Convention and Articles 1–14 of the Paris Convention, were enumerated instead.⁹² A similar text was agreed to by the NEA Group at its June 1987 meeting, except that Article II became Article III as a result of the already mentioned insertion of a new provision specifically dealing with the mutual extension of the "geographical scope" of the operator's liability.⁹³ Article III was then provisionally adopted as Article IV at the Joint IAEA/NEA Working Group meeting in October 1987, as a result of a reordering of articles within the 1987 Draft Joint Protocol,⁹⁴ and thereafter remained unchanged until its final adoption at the Diplomatic Conference in 1988.

The text of what became Article IV of the Joint Protocol was presented to the October 1987 Joint IAEA/NEA Working Group meeting with a commentary stating that the choice of a positive enumeration of the applicable articles was:

"mainly determined by practical considerations as well as by the wish not to mention expressly those Articles of the Conventions which are not directly relevant to the concept of non-Contracting States and to exclude expressly not only the procedural Articles but also those Articles of either Convention which have no counterpart in the other (Articles 7(e) and 17 [of the Paris Convention], XVI [of the Vienna Convention]) or are different in substance (Article 6(e) [of the Paris Convention])."

The commentary also "submitted that the positive formula is as valid as the negative one; it may even be preferable as it does not state exceptions" but further noted that, "contrary to the 1974 Draft Joint Protocol, Articles 6(e) and 7(e) of the [Paris Convention] are not excluded".⁹⁵

This latter notation makes it clear that there is no exact correspondence between the articles in either Convention whose application was expressly excluded by Article I of the 1974 Draft Joint Protocol and the articles whose application is now implicitly excluded by Article IV of the Joint Protocol, and that the change from a negative to a positive enumeration was largely but not exclusively a drafting change. In particular, the fact that Articles $6(e)^{96}$ and $7(e)^{97}$ of the Paris Convention are no longer excluded entails that there is no perfect reciprocity

⁹⁷ Article 7(e) of the Paris Convention, which is mentioned in Section 1.1, relates to transit of nuclear material and allows a Contracting Party to subject the transit of nuclear substances through its territory to the condition that the maximum liability of the operator of the foreign nuclear installation be increased up to the maximum liability applicable within its own territory. No corresponding provision exists in either the original or the 1997 Vienna Convention.

⁹² IAEA document 250-N5.TC-462.6, Annex C.

⁹³ OECD-NEA document LEG/DOC(87)3 (Rev. 1).

⁹⁴ IAEA document N5-TC-643.

⁹⁵ IAEA document N5/TC/643(3), p. 5.

⁹⁶ Article 6(e) of the Paris Convention relates to subrogation in cases where persons other than the operator of a nuclear installation have paid compensation for damage caused by a nuclear incident. Under Article 6(d) of the Paris Convention, "any person" who has paid compensation pursuant to any international transport agreement or to the legislation of a non-Contracting State acquires, up to the amount which he has paid, the rights under the Paris Convention of the person whom he has so compensated. Article IX.2(a) of the Vienna Convention restricts such subrogation to the nationals of the Contracting Parties. Article 6(e) of the Paris Convention, which has no corresponding provision in the Vienna Convention, also gives a right of subrogation to persons with a principal base of business in a Contracting State, where they or their servants pay compensation for damage suffered in a non-Contracting State or caused by a nuclear incident occurring in such State.

in the substantive obligations undertaken by the Parties to the two Conventions under Article IV of the Joint Protocol.⁹⁸

Another difference between Article I of the original 1974 Draft Joint Protocol and Article IV of the Joint Protocol relates to the terms whereby the same idea is expressed and does not appear to entail a major difference of substance. Whereas Article I of the 1974 Draft provided that the Parties to the Protocol which were Parties to either the Vienna Convention or the Paris Convention would be "considered as if they were Parties" to the other Convention (with the exception of the enumerated articles), Article IV of the Joint Protocol now provides that (the enumerated articles of) either the Vienna Convention or the Paris Convention "shall be applied" with respect to the Parties to the Protocol which are Parties to the other Convention "in the same manner" as between Parties to that Convention. As was explained in the documentation examined by the Joint IAEA/NEA Working Group meeting of October 1987, the revised wording "aims at establishing equal treatment as regards the operative Articles of either Convention without implying the status of a full Contracting Party", thus addressing "the concern that the wording used in Article I of the 1974 draft...might be too far-reaching in the light of international treaty practice".⁹⁹ But the idea that the Joint Protocol would not afford to the Parties of either Convention the full status of a Party to the other was clear to the drafters from the beginning.¹⁰⁰ Moreover, as has been pointed out, the 1987 Explanatory Note, while confirming that the new wording in Article IV is intended to make it clear that the Joint Protocol "does not afford the full status of a Party to the other Convention",¹⁰¹ still emphasizes that the principle underlying that provision "is to create a link or 'bridge' between the two Conventions by abolishing the distinction between Contracting Parties and non-Contracting States as regards the operative provisions of either Convention".¹⁰²

Inasmuch as the mutual extension of the operator's liability under either Convention to damage suffered in the territory of Contracting Parties to the other Convention is now specifically provided for in Article II of the Joint Protocol, the main effects of Article IV of the Joint Protocol may be said to concern, in respect of the transport of nuclear material: the determination of the liable operator; and the determination of the State whose courts have jurisdiction. Moreover, a more general issue that does not exclusively relate to transport cases may arise in respect of

Article 7(e) of the Paris Convention will be dealt with in greater detail when discussing the amount of compensation to be paid under the applicable Convention, in Section 3.2.3.

⁹⁸ The following comments were made in the documentation presented to the October 1987 Joint IAEA/NEA Working Group meeting in respect of this choice:

[&]quot;As a matter of fact, Article 6(e) is confined to compensation in respect of damage caused by a nuclear incident occurring in the territory of a non-Contracting State or in respect of damage caused in such territory. This rule would remain unaffected if an operator is liable under the [Paris Convention], but would not apply to incidents occurring and damage suffered in Contracting States to the [Vienna Convention] as they are not considered as non-Contracting States under the terms of draft Article III [corresponding to Article IV of the Joint Protocol]....As regards Article 7(e) of the [Paris Convention], it remains applicable among its Contracting Parties. As the Joint Protocol establishes the principle of equal treatment and non-discrimination between the Contracting Parties to either Convention, this Article should also apply in the relationship between those Parties. It is to be noted that this provision would be applied among Contracting Parties to the [Paris Convention] only, if a [Paris Convention operator] assumes liability by contract during carriage of nuclear material between his installation and that of a [Vienna Convention operator] (Articles 4(a)(i) and (b)(i) [of the Paris Convention], II.1(b)(i) and (c)(i) [of the Vienna Convention], in connection with Article II above [corresponding to Article III of the Joint Protocol]). This assumption of liability would preserve the application of the [Paris Convention] and hence of the [Brussels Convention]" (IAEA document N5/TC/643(3), pp. 5–6).

⁹⁹ IAEA document N5/TC/643(3), p. 6.

¹⁰⁰ For example, WD I and WD II point out that the original formula whereby the Parties to the Vienna Convention would be "treated as if they were" Parties to the Paris Convention would imply that they would not, as a result thereof, be eligible for accession to the Brussels Convention, which, under Article 19 thereof, is only open to Contracting Parties to the Paris Convention.

¹⁰¹ 1987 Explanatory Note, para. 9.

¹⁰² 1987 Explanatory Note, para. 4.

the amount of compensation to be paid. There are, of course other effects,¹⁰³ but these three were singled out from the beginning as the major effects of the "formula under which the parties to one Convention are treated as if they were parties to the other Convention"¹⁰⁴ and appear to deserve some comment.

The determination of the liable operator 3.2.1.

As is pointed out in Section 1.2, the general rule under both the Paris Convention and the Vienna Convention is that, in the case of a nuclear accident occurring during transport of nuclear material between the operators of two nuclear installations, the sending operator is liable until the receiving operator has assumed liability pursuant to the express terms of a written contract or, in the absence of such contract, until the receiving operator has taken charge of the materials involved.¹⁰⁵ However, this general rule only applies if both operators are within the territory of Contracting Parties to the same Convention: if the nuclear materials are sent to a person in a non-Contracting State, including a Contracting Party to a different Convention, the sending operator remains liable until the nuclear materials have been unloaded from the means of transport by which they arrived in the territory of that State; conversely, if the nuclear materials are sent from a person in a non-Contracting State, including a Contracting Party to a different Convention, liability is imposed upon the receiving operator from the moment when the materials have been loaded on the means of transport.¹⁰⁶ As a result, if a nuclear incident occurs in the course of transport of nuclear material between operators of nuclear installations situated in the territories of States party to different nuclear liability Conventions, the transfer of liability between the sending and the receiving operator cannot take place on the basis of a written contract between them or when one of them has taken charge of the materials involved, and both operators may be held liable, each under the applicable Convention, for the entire time when the nuclear materials are on the means of transport.

The effect of Article IV of the Joint Protocol is that the transport of nuclear material between the operators of two nuclear installations situated within the territory of Contracting Parties to different Conventions where both are party to the Joint Protocol is treated as if the transport took place between operators of installations situated in the territory of Contracting Parties to the same Convention. Consequently, as is stated in the "Background Material" presented at the Joint IAEA/NEA Working Group meeting in October 1987, "the transfer of liability between [Vienna Convention and Paris Convention] operators is determined by the terms of a contract in writing or, in the absence thereof, by taking charge of the nuclear material".¹⁰⁷

From the perspective of the Contracting Parties to the Paris Convention, the possibility for a Paris Convention-Joint Protocol operator to assume liability by contract in the case of the transport of nuclear material between his installation and an installation situated in the territory of a Contracting Party to the Vienna Convention and the Joint Protocol is especially important since, under Article III of the Joint Protocol, which is examined in Section 3.3, it would

- (b) the transport of nuclear material (Articles 4(a)(iv) and (b)(iv) [Paris Convention], II.1(b)(iv) and (c)(iv) [Vienna Convention]):
- (c) the right of subrogation (Articles 6(d) and (e) [Paris Convention], IX.2 [Vienna Convention]);
- (d) the free transfer of compensation and funds provided by insurance or other financial security (Articles 12 [Paris Convention] and XV [Vienna Convention]);
- (e) the jurisdictional provisions (Articles 13(a) to (c) [Paris Convention], XI [Vienna Convention]);
- (f) the enforcement of judgments (Articles 13(d) [Paris Convention], XII [Vienna Convention]);
- (g) the principle of non-discrimination (Articles 14 [Paris Convention], XIII [Vienna Convention])" (IAEA document N5/TC/ 643(4), para. 36).

¹⁰⁵ Article 4 (a)(i)–(iii) and (b)(i)–(iii) of the Paris Convention and Article II.1 (b)(i)–(iii) and (c) (i)–(iii) of the Vienna Convention.

¹⁰⁷ IAEA document N5/TC/643(4), para. 38.

¹⁰³ In the "Background Material" presented to the October 1987 Joint IAEA/NEA Working Group meeting, it was stated that: "the distinction between 'Contracting Parties' or 'Installation State' (the latter term being used by the [Vienna Convention] only but defined with reference to 'Contracting Party' in Article I.1(d)) and non-Contracting States is of particular significance in both

Conventions in respect to: (a) their geographical scope (Article 2 [Paris Convention]);

¹⁰⁴ WD II, para. 52.

¹⁰⁶ Article 4 (a)(iv) and (b)(iv) of the Paris Convention and Article II.1 (b)(iv) and (c)(iv) of the Vienna Convention.

preserve the application of the Paris Convention and, therefore, of the Brussels Convention.¹⁰⁸ However, even where a Paris Convention–Brussels Convention–Joint Protocol operator has assumed liability by contract in the case of the transport of nuclear material between his installation and an installation situated in the territory of a Vienna Convention–Joint Protocol State, the Brussels Convention, under Article 2 thereof, would only apply provided that the incident does not take place wholly in a non-Contracting State and that the courts of a Contracting Party have jurisdiction under the Paris Convention. Moreover, the additional funds available under the Brussels Convention could only be used to compensate damage suffered in the Contracting Parties (or on the high seas on board a ship or aircraft registered in a Contracting Party or by a national of a Contracting Party).

3.2.2. The determination of the State whose courts have jurisdiction

As is pointed out in Section 1.2, the general rule under all nuclear liability conventions is that jurisdiction lies with the courts of the Incident State, that is, the Contracting Party within whose territory, including maritime areas subject to its full sovereignty, the nuclear incident occurs.¹⁰⁹ Under the 1997 Vienna Protocol (and under the Paris Convention once it is amended by the 2004 Protocol), the courts of that Contracting Party have jurisdiction even if the incident occurs beyond its territorial sea, within the area of its exclusive economic zone (or, if such a zone has not been established, within an area not exceeding the limits thereof), provided that it has notified the depository of such area prior to the nuclear incident.¹¹⁰ On the other hand, if the nuclear incident occurs outside the territory of any Contracting Party or, under the 1997 Vienna Protocol (or the Paris Convention once it is amended by the 2004 Protocol), outside the area of the exclusive economic zone of any Contracting Party (e.g. on the high seas), jurisdiction lies with the courts of the Installation State, that is, the Contracting Party within whose territory the nuclear installation of the operator liable is situated.¹¹¹

As a result, no particular problem would arise in the case of a nuclear incident at a nuclear installation, since only the Convention to which the Installation State is a Contracting Party would apply and the courts of that State, which is at the same time the Installation State and the Incident State, would have exclusive jurisdiction thereunder. On the other hand, in the case of a nuclear incident involving the transport of nuclear materials, problems may arise if the transport takes place between operators situated in Contracting Parties to different nuclear liability Conventions, each of which is to be considered as a non-Contracting State vis-à-vis the other: as is pointed out in Section 3.2.1, both the sending and the receiving operators may be held liable, each of them under the applicable Convention, and in that case, the courts in both States, both being the Installation State under the applicable Convention, would have jurisdiction; on the other hand, the Contracting Parties to each of the Conventions would be under no specific obligation to ensure that final

¹⁰⁸ The problems of the possible effects of the Joint Protocol on the Brussels Convention were on the minds of the drafters from the very beginning: in WD II (para. 55), it was pointed out that one effect of the Protocol would be that the Brussels Convention (at the time not yet in force) would not be applicable in some cases where it would have been applicable (had it been in force) without the Protocol. It was pointed out, for example, that, in the case of the transport of nuclear material between a Paris Convention–Brussels Convention–Joint Protocol operator and a Vienna Convention–Joint Protocol operator, had an incident occurred in the Paris Convention–Brussels Convention would not have applied, whereas it would have applied without the Protocol. According to WD II:

[&]quot;a possible solution to this problem may be that all [Paris Convention] States ensure, either in their domestic legislation or by some administrative means, that operators of installations in their territory assume liability by contract for any incidents which may occur in the course of carriage between their installation and a [Vienna Convention] State and for which the system of the [Brussels Convention] would be applicable".

After the Joint Protocol was adopted, a solution along those lines was recommended to the Contracting Parties to the Paris Convention by the OECD Council in 1989.

¹⁰⁹ Article 13(a) of the Paris Convention and Article XI.1 of the Vienna Convention.

¹¹⁰ Articles XI.1 bis of the 1997 Vienna Convention and Article 13(b) of the Paris Convention as amended by the 2004 Protocol.

¹¹¹ Article 13(b) of the Paris Convention and Article XI.2 of the Vienna Convention; Article XI.2 1997 of the Vienna Convention; Article 13(c) of the Paris Convention as amended by the 2004 Protocol.

judgements entered by a court in a Contracting Party to the other Convention are recognized and enforced within their territories.

Under Article IV of the Joint Protocol, the jurisdictional provisions in either the Vienna Convention or the Paris Convention are to be applied with respect to the Contracting Parties to the other Convention and the Joint Protocol "in the same manner as between Parties" to that Convention. Therefore, in the case of a nuclear incident involving the transport of nuclear materials between operators whose installations are situated in Contracting Parties to different nuclear liability Conventions but which are both party to the Joint Protocol, jurisdiction lies with the courts of the Incident State (including, in the case of a Party to the 1997 Vienna Protocol or, in the future, to the Paris Convention as amended by the 2004 Protocol, in cases where the nuclear incident occurs within the area of the Incident State's exclusive economic zone). On the other hand, if the operator is liable under the Convention to which the Installation State is a Party, the competent court in the Incident State will still have to refer, in principle, to that Convention under Article III of the Joint Protocol.¹¹²

The above interpretation — whereby in the event of an accident during transport involving operators in States that are Contracting Parties to different Conventions (the Paris Convention or the Vienna Convention) but which are both party to the Joint Protocol, the rules giving competence to the courts of the Incident State should apply — appears to be the one most in accordance with the ordinary meaning to be given to the terms of Article IV of the Joint Protocol in their context and in the light of the Joint Protocol's object and purpose.¹¹³ As pointed out in its Preamble, the object and purpose of the Joint Protocol is precisely to "establish a link" between the Paris Convention and the Vienna Convention, and the 1987 Explanatory Note confirms that this link was provided by "abolishing the distinction between Contracting Parties and non-Contracting States as regards the operative provisions of either Convention".¹¹⁴ Article IV of the Joint Protocol expressly enumerates the provisions relating to jurisdiction in both the Vienna Convention and the Paris Convention among those that have to be applied "in the same manner as between Parties": there would be no need for such enumeration if those provisions were still to apply as between non-Contracting States. As is pointed out in Section 3.3, as far as the context is concerned, Article III of the Joint Protocol, relating to the choice of the applicable Convention, is based on the assumption that the applicable Convention is the one in force for the Installation State, even where jurisdiction lies with the courts of a Contracting Party to another Convention: there would be no need for such a provision if, under the Joint Protocol, jurisdiction in transport cases involving Parties to different Conventions lay always with the courts of the Installation State.

The same interpretation is also supported by the travaux préparatoires.¹¹⁵ A clear example was given in this respect in WD II:

"If...an operator in State A, which has ratified the Paris Convention and the Protocol(s) sends nuclear material from his nuclear installation to an operator in State B, which has ratified the Vienna Convention and the Protocol(s), and an incident occurs in State B caused by that material, the legal situation would be as follows...:

— Article 4(c)(ii) of the Paris Convention and Article II.1(c)(iv) of the Vienna Convention would not apply as A and B would not be considered as non-contracting States within the meaning of these provisions. The liable operator would be determined pursuant to the identical provisions of Article 4(a)(i) or (ii) of the Paris Convention and Article II.1(c)(i) or (ii) of the Vienna Convention;

¹¹² See Section 3.3.

¹¹³ Under Article 31.1 of the 1969 Vienna Convention on the Law of Treaties: "A treaty shall be interpreted in good faith in accordance with the ordinary meaning to be given to the terms of the treaty in their context and in the light of its object and purpose".
¹¹⁴ 1987 Explanatory Note, para. 4.

¹¹⁵ Under Article 32 of the 1969 Vienna Convention on the Law of Treaties:

[&]quot;Recourse may be had to supplementary means of interpretation, including the preparatory work of the treaty and the circumstances of its conclusion, in order to confirm the meaning resulting from the application of article 31, or to determine the meaning when the interpretation according to article 31: (a) leaves the meaning ambiguous or obscure; or (b) leads to a result which is manifestly absurd or unreasonable."

- Jurisdiction would lie with the courts of B under both Article 13(a) of the Paris Convention and Article XI.1 of the Vienna Convention (the first alternatives of Article 13(b) of the Paris Convention and XI.2 of the Vienna Convention not being applicable under the terms of the Protocol(s));
- The amount of compensation to be paid would be governed by the law of the Installation State...."¹¹⁶

Of course, WD II was more directly relevant for the adoption of Article I of the original 1974 Draft Joint Protocol. But the same example was then repeated almost verbatim in the NEA Secretariat's Note presented to the June 1987 NEA Group meeting,¹¹⁷ and in the "Background Material" presented to the October 1987 Joint IAEA/NEA Working Group meeting,¹¹⁸ both of which were at the basis of the definitive adoption of Article IV of the Joint Protocol.

A different interpretation was recently put forward by a leading nuclear law expert on the occasion of a 2005 NEA Workshop. According to this interpretation, in the event of an accident during transport involving operators in Contracting Parties to different Conventions (the Paris Convention or the Vienna Convention) but which are both party to the Joint Protocol, jurisdiction should still lie with the courts of the Installation State rather than those of the Incident State.¹¹⁹ However, the interpretation given in this Explanatory Text was confirmed by the NEA Secretariat on the same occasion.¹²⁰

3.2.3. The amount of compensation to be paid

As is pointed out in Section 3.2.2, the amount of compensation is governed by the law of the Installation State both in the case where jurisdiction lies with the courts of that same State and in the case where jurisdiction lies with the courts of the Incident State, if different from the Installation State. This clearly results from Article III of the Joint Protocol, which relates to the choice of the applicable nuclear liability Convention and is examined in Section 3.3. But even in the absence of such a provision, Article IV of the Joint Convention would still have led to the same result, inasmuch as both Article V of the Vienna Convention and Article 7 of the Paris Convention refer to the Installation State as the State competent to determine, within the limits allowed by either Convention, the amount of the operator's liability. The same goes for the amount that has to be covered by insurance or other financial security under Article VII of the Vienna Convention.

One issue more directly related to the effects of Article III of the Joint Protocol relates to reciprocity in respect of the amount of the operator's liability, since, as is pointed out in Section 1.1, the nuclear liability Conventions differ

¹²⁰ "Note by the OECD/NEA Secretariat", *Indemnification of Damage in the Event of a Nuclear Accident (Proc. Workshop, Bratislava, 2005)*, p. 101 ff. The Note concluded that:

"in respect of an accident involving only [Paris Convention/Vienna Convention] countries that are also Party to the [Joint Protocol], the rules giving competence to the courts of the accident State should apply. On the other hand, to the extent that non-[Joint Protocol] States would be involved because of damage suffered in their respective territories, there may be a problem if they object to the designation of the competent court under the rules of the [Joint Protocol]" (p. 102).

This latter observation hints at the conflicts of treaty obligations that may arise for Parties to the Joint Protocol vis-à-vis Parties to either the Vienna Convention or the Paris Convention which are not Parties to the Joint Protocol. Such conflicts may arise as a result of the decision to allow the Joint Protocol to enter into force even if not all Parties to the Vienna Convention and the Paris Convention have ratified it (see Section 4.2). It seems sufficient here to point out that conflicts of that nature may not only arise in respect of the jurisdictional provisions of the Vienna Convention and the Paris Convention, but also in respect of other provisions thereof (e.g. those relating to the determination of the liable operator).

¹¹⁶ WD II, para. 52.

¹¹⁷ OECD-NEA document LEG/DOC(87)3 (Rev. 1), para. 38.

¹¹⁸ IAEA document N5/TC/643(4), para. 38.

¹¹⁹ Pelzer, N., "Interpretation of the Joint Protocol in transport cases — The German position", *Indemnification of Damage in the Event of a Nuclear Accident (Proc. Workshop, Bratislava, 2005)*, OECD/NEA, Paris (2006) p. 105 ff. According to this view, the opposite interpretation would be in contradiction to Article III.3 of the Joint Protocol, which provides for the application of the Convention to which the Installation State is a Party; moreover, it would entail that the courts of two States would have jurisdiction, since the courts of the Installation State would still have jurisdiction under either the Paris Convention or the Vienna Convention "in relation to" States party to that Convention that were not party to the Joint Protocol.

considerably as to that amount. More specifically, under Article V of the 1963 Vienna Convention, the minimum liability amount is \$5 million,¹²¹ whereas Article 7 of the Paris Convention, as at present in force, establishes a maximum amount of 15 million SDRs¹²² but then allows the Installation State to establish by legislation a greater or lesser amount, taking into account the possibilities for the operator to obtain the required insurance or other financial security. Under Article V of the 1997 Vienna Convention, the minimum amount is raised to 300 million SDRs¹²³ (or 150 million SDRs, provided that public funds are made available by the Installation State to compensate the damage in excess, up to 300 million SDRs), but Article 7 of the Paris Convention, as it will be amended by the 2004 Protocol, will also establish a minimum, rather than a maximum, amount and will raise that amount to €700 million.

As far as the Contracting Parties to the Paris Convention are concerned, a partial remedy to the potential imbalance in the liability amounts in a transport situation is represented by Article 7(e), which, as mentioned, allows a Contracting Party to subject the transit of nuclear material through its territory to the condition that the maximum amount of liability of the foreign operator concerned be increased to the maximum amount of liability of operators of nuclear installations situated in its own territory. As also mentioned, as a result of the drafting changes which led from Article I of the 1974 Draft Joint Protocol to Article IV of the Joint Protocol, a Contracting Party to both the Paris Convention and the Joint Protocol can avail itself of this provision not only in respect of other Contracting Parties to the Paris Convention, but also in respect of the Contracting Parties to both the Vienna Convention, since there is no express provision corresponding to Article 7(e) of the Paris Convention in either the 1963 Vienna Convention or the 1997 Vienna Convention. On the other hand, given the much higher amount of liability available under the Paris Convention, it is unlikely that this situation would in practice pose a problem for the Contracting Parties to the Vienna Convention.

Quite apart from the possibility envisaged in Article 7(e) of the Paris Convention, the more general question arises as to whether, under the terms of the Joint Protocol, the Contracting Parties to one Convention may limit the amount of compensation available under their legislation to the lesser amount that may be available under the legislation of the Contracting Parties to the other Convention. The drafters of the Joint Protocol were aware of this question from the very beginning, and the answer they gave in the earliest stages of the preparatory work was as follows:

"As far as the relationship between the Parties to the same Convention [is] concerned such restriction is not possible (Article 7(d) of the Paris Convention). Consequently, if there is treatment as Party in the general sense...the same would apply. Such limitations would also be contrary to the non-discrimination Articles of both Conventions (Article 14(a) of the Paris Convention and Article XIII of the Vienna Convention) and would put nationals of non-parties to both Conventions who suffer damage in the territory of the Parties or on the high seas in a more favourable position than nationals of the Parties".¹²⁴

These words were repeated verbatim in WD II,¹²⁵ but not in the Note presented to the June 1987 NEA Group meeting, nor in the "Background Material" presented to the October 1987 Joint IAEA/NEA Working Group meeting, which were at the basis of the definitive adoption of Article IV of the Joint Protocol. On the other hand, these documents did point out that, as a result of the approach taken in the Joint Protocol, "The maximum amount of the operator's liability as fixed by his Installation State's legislation pursuant to the Convention to which the latter is a

¹²¹ Article V. 2 of the Vienna Convention defines the US dollar referred to in the Convention as "a unit of account equivalent to the value of the United States dollar in terms of gold on 29 April 1963, that is to say US \$35 per one troy ounce of fine gold".

¹²² SDR is the acronym for 'special drawing right', which is defined in Article 7 of the Paris Convention as "the unit of account defined by the International Monetary Fund and used by it for its own operations and transactions".

¹²³ Like Article 7 of the Paris Convention, Article I.1(p) of the 1997 Vienna Convention uses SDR as the acronym for 'special drawing right', defined as "the unit of account defined by the International Monetary Fund and used by it for its own operations and transactions".

¹²⁴ WD I, para. III.3.

¹²⁵ WD II, paras 61–63.

Party, covers nuclear damage suffered in [Vienna Convention] as well as in [Paris Convention] States without discrimination".¹²⁶

Even if the interpretation given above is considered to be in line with both the Paris Convention and the Vienna Convention as they existed at the time the Joint Protocol was adopted, the additional question arises as to whether, and to what extent, the situation changed after the entry into force of the 1997 Vienna Protocol and/or will change after the Paris Convention is further amended by the 2004 Protocol.

As far as the Vienna Convention is concerned, it could be argued that the situation has indeed changed as a result of the entry into force of the 1997 Vienna Protocol. In fact, a Contracting Party to the Vienna Protocol may avail itself of Article XIII.2 of the 1997 Vienna Convention, providing that:

"notwithstanding paragraph 1 of this Article [relating to non-discrimination], insofar as compensation for nuclear damage is in excess of 150 million SDRs, the legislation of the Installation State may derogate from the provisions of this Convention with respect to nuclear damage suffered in the territory, or in any maritime zone established in accordance with the international law of the sea, of another State which, at the time of the incident, has a nuclear installation in such territory, to the extent that it does not afford reciprocal benefits of an equivalent amount."

To some extent, this provision was inspired by Article 15(b) of the Paris Convention, which is referred to below; however, unlike Article 15(b) of the Paris Convention, Article XIII.2 of the 1997 Vienna Convention appears to apply to compensation in excess of 150 million SDRs quite irrespective of the basis for such compensation, for example, even if such compensation is based on the operator's liability and not on public funds in excess thereof. The Explanatory Text relating to the Vienna Protocol points out that the interpretation of this provision is not very easy:

"[W]hereas the context of the new provision might be seen as implying that a derogation could only relate to the non-discrimination principle, a broader interpretation is also possible, since reference is made to the possibility of derogating from 'the provisions' of the Convention. Under this broader interpretation, it could be argued, for example, that the legislation of the Installation State could establish lower amounts of compensation in respect of damage suffered in a State not affording reciprocal benefits; this interpretation may be seen as reinforced by the fact that the new provision is intended to apply to a nuclear State not affording 'reciprocal benefits of an equivalent amount', a language which clearly refers to the amounts of compensation''.¹²⁷

Inasmuch as the new Article XIII.2 of the 1997 Vienna Convention refers to "another State", it is clearly intended to apply both to the other Contracting Parties to the Vienna Protocol and to non-Contracting States, to the extent that damage suffered in such States is covered. In any event, if a Contracting Party to the Vienna Protocol ratifies or accedes to the Joint Protocol, it would have to cover damage suffered in the Contracting Parties to both the Paris Convention and the Joint Protocol, but Article IV of the Joint Protocol would allow it to apply the new Article XIII.2 to these latter States "in the same manner" as it applies between Contracting Parties to the Vienna Protocol.

Similar considerations may be said to apply to the Paris Convention once it is amended by the 2004 Protocol. The new Article 2 of the Paris Convention will make it explicit that, as far as Contracting Parties to the Paris Convention which are also Contracting Parties to the Joint Protocol are concerned, damage suffered in Contracting Parties to the Vienna Convention which are in their turn also Contracting Parties to the Joint Protocol will be covered under the amended Paris Convention, whereas the amended Paris Convention will only cover damage suffered in nuclear non-Contracting States which are not party to the Joint Protocol if these States, at the time of the nuclear incident, have nuclear liability legislation in place which affords "equivalent reciprocal benefits". This provision is in line with Article II of the Joint Protocol. The new Article 3(g) of the Paris Convention will allow a Contracting Party, in cases where the amended Paris Convention applies to a non-Contracting State, "to establish in respect of nuclear damage amounts of

¹²⁶ OECD-NEA document LEG/DOC(87)3 (Rev. 1), para. 38; IAEA document N5/TC/643(4), para. 38.

¹²⁷ IAEA International Law Series No. 3, Section 2.8.
liability lower than the minimum amounts established under this Article...to the extent that such State does not afford reciprocal benefits of an equivalent amount", but this provision explicitly confines its application to those non-Contracting States which are not Contracting Parties to both the Vienna Convention and the Joint Protocol.

On the other hand, attention has to be focused on the new text of Article 15(b) of the Paris Convention. At the time when the Joint Protocol was adopted, no attention was paid to this provision, which, as at present in force, provides that, "in so far as compensation for damage involves public funds" and is in excess of the minimum amount of the operator's liability provided for in Article 7 of the Paris Convention, "any such measure in whatever form may be applied under conditions which may derogate from the provisions of this Convention". In contrast, the new Article 15(b), as it will be amended by the 2004 Protocol, no longer refers to public funds and simply provides, in language similar to the new Article XIII.2 of the 1997 Vienna Convention, that, "Insofar as compensation for nuclear damage is in excess of the 700 million euros referred to in Article 7(a), any such measure in whatever form may be applied under conditions which may derogate from provisions of this Convention". Inasmuch as this provision will apply as between Contracting Parties to the amended Paris Convention, the effect of Article IV of the Joint Protocol is that a Contracting Party to both the amended Paris Convention and the Joint Protocol will be allowed to apply that provision "in the same manner" to the Contracting Parties to both the Vienna Convention and the Joint Protocol.

It appears, therefore, that, under the above interpretation of Article XIII, para. 2, of the 1997 Vienna Convention, the Contracting Parties to the Vienna Protocol may resort to reciprocity in respect of compensation available in excess of 150 million SDRs, but, for those of them that are party to the Joint Protocol, it is highly unlikely that they may in fact do so vis-à-vis the Contracting Parties to the Paris Convention that are also party to the Joint Protocol, in view of the much higher amount of compensation that will be available under this latter Convention once it is amended by the 2004 Protocol. On the other hand, despite the much lower amount of compensation available under the other Convention, the Contracting Parties to the amended Paris Convention that are also party to the Joint Protocol will only be allowed to resort to reciprocity vis-à-vis the Contracting Parties to the Vienna Convention in respect of compensation available in excess of \in 700 million.¹²⁸

3.3. THE CHOICE OF THE APPLICABLE CONVENTION (ARTICLE III)

As is pointed out in Section 2.2, the Joint Protocol aims not only at establishing a treaty link between the Contracting Parties to the Vienna Convention and the Contracting Parties to the Paris Convention, but also at eliminating problems arising from the simultaneous application of both Conventions to the same nuclear incident. Thus, the second principle underlining the Joint Protocol consists in making either the Vienna Convention or the Paris Convention or the Paris Convention exclusively apply to a nuclear incident by means of an appropriate conflict rule.

At the time when the Joint Protocol was originally drafted, it was felt that "the formula under which the parties to one Convention are treated as if they were parties to the other Convention would achieve, at the same time, the objective of avoiding conflicts between and simultaneous application of the two Conventions as well as establishing a relationship between them".¹²⁹ However, it was also felt that, although that formula would result in a clear "determination of the operator liable, the competent court and the national law under which the compensation is to be paid", it might still be necessary to determine the Convention applicable: it was pointed out, in particular, that the

¹²⁸ This interpretation is based exclusively on the text of the Paris Convention as amended by the 2004 Protocol. However, mention must be made of a Joint Declaration made on 23 November 2004 by the Signatories to the Protocol of 12 February 2004 to Amend the Paris Convention, whereby:

[&]quot;if the following reservation is made in accordance with Article 18 of the Paris Convention, such a reservation is accepted:

^{&#}x27;[Name of the State making the reservation], without prejudice to Article 2(a)(iii), reserves the right to establish in respect of nuclear damage suffered in the territory of, or in any maritime zones established in accordance with international law of, or on board a ship or aircraft registered by, a State other than [name of the State making the reservation], amounts of liability lower than the minimum amount established under Article 7(a) to the extent that such other State does not afford reciprocal benefits of an equivalent amount'''.

¹²⁹ WD II, para. 52.

"Brussels Supplementary Convention presupposes, inter alia, that the operator is liable under the Paris Convention" and that "the Conventions have different rules on the right of subrogation and recourse". It was further considered in this respect that "the general formula that the Parties to one Convention shall be treated as if they were Parties to the other Convention would not seem to provide a clear determination of the Convention applicable", since "the identical provisions of both Conventions merely lead to the same result". Moreover, "Articles 11 of the Paris Convention and VIII of the Vienna Convention do not appear to be relevant in this respect, as they deal only with the nature, form and extent of the compensation as to be settled by the law of the competent court and presuppose the applicability of the Convention in question".¹³⁰

Once it was established that an appropriate conflict rule was required in order to determine the applicable Convention, WD II pointed out that the choice could be made in two ways: it "could be either the Convention to which the Installation State of the operator liable is a Party or the Convention to which the State whose courts have jurisdiction is a Party."¹³¹ However, the first solution was the one preferred from the very beginning and consequently was already reflected in the 1974 Draft Joint Protocol (Article II). WD II explained this choice by pointing out that, under the alternative solution:

"the liable operator may be held liable under a Convention to which his Installation State is not a Party. For example, as the provisions of the Paris Convention on the rights of subrogation and recourse are wider than those of the Vienna Convention, the Parties to the latter would have to amend their national law to provide for the case that an action is brought before a [Paris Convention] State court against a [Vienna Convention] State operator under Article 6(d) or (e) of the Paris Convention. Such legislation would, however, not be in conformity with the Vienna Convention."¹³²

Under the preferred solution, no such problems would arise:

"the operator would always be held liable under the Convention which corresponds to his own national law. The competent court may have to apply a Convention different from the lex fori, but that is not unusual in conflict of law cases".¹³³

¹³⁰ WD II, para. 53.

¹³¹ WD II, para. 54.

¹³² WD II, para. 54.

¹³³ WD II, para. 54. At the 2005 Bratislava Workshop, the interpretation whereby the competent court would have to apply the Convention to which the Installation State is a Party rather than the Convention in force for its own State was criticized by some nuclear law experts as not "reasonable" (N. Pelzer) and as "unacceptable in terms of constitutionality" (V. Lamm) (see Indemnification of Damage in the Event of a Nuclear Accident (Proc. Workshop, Bratislava, 2005), pp. 106 and 109, respectively). In view of these late criticisms, it may be of interest to recall that, although para. 39 of the original Note presented to the June 1987 NEA Group meeting (OECD-NEA document LEG/DOC(87)3) also referred to the fact that, under the preferred solution, the competent court may have to apply "a Convention and national law different from the lex fori" (just as in para. 54 of WD II, which is quoted in the text), during the discussion regarding the determination of the applicable Convention at the June 1987 meeting, that language was criticized and eventually changed: "several Representatives considered that it was incorrect to refer to a foreign convention since, if the Protocol were ratified, the Convention became an integral part of national law", and the Group eventually agreed to delete the word "Convention" and refer only to "national law different from the lex fori", as suggested by the Swedish representative (OECD-NEA document SEN/LEG(87)3, para. 29). Consequently, the revised Note (OECD-NEA document LEG/DOC(87)3 (Rev. 1), para. 39), as well as the "Background Material" presented to the October 1987 Joint IAEA/NEA Working Group meeting (IAEA document N5/TC/643(4), para. 39), referred to the fact that "the competent court may have to apply a national law different from the lex fori". In fact, from the point of view of public international law, the obligation for the Incident State to apply the Convention to which the Installation State is a Party would not derive from this latter Convention, which is not in force for the Incident State, but rather from the Joint Protocol, a treaty to which the Incident State itself is a Party; from the point of view of private international law, the competent court in the Incident State would apply the Convention to which the Installation State is a Party, as part of the national law of this latter State, by virtue of a conflict rule inserted in the lex fori by the Joint Protocol, either as directly applicable or as implemented in the lex fori.

WD II then points out that:

"Moreover, the application of the foreign Convention will in most cases be limited to the amount of compensation available under the foreign operators' national law, while the nature, form and extent of the compensation as well as the equitable distribution thereof will be governed by the national law of the competent court."¹³⁴

These reasons were repeated almost verbatim in the Note presented to the June 1987 NEA Group meeting¹³⁵ and in the "Background Material" presented to the October 1987 Joint IAEA/NEA Working Group meeting,¹³⁶ which were the basis for the adoption of Article III of the Joint Protocol.

The conflict rule determining the applicable Convention was formulated in Article II of the original 1974 Draft Joint Protocol by simply referring to "the Convention to which the Installation State of the operator liable, by virtue of either Convention, is liable".¹³⁷ The substance of this rule has not changed, despite the drafting changes to which it was subjected to before its final adoption. It was felt, in particular, that, inasmuch as both Conventions apply not only to nuclear incidents occurring in nuclear installations, but also to nuclear incidents occurring during carriage of nuclear materials, two conflict rules (as opposed to one) were required.¹³⁸ Thus, in the Note presented to the June 1987 NEA Group meeting the NEA Secretariat proposed a revised text for Article II,¹³⁹ which was then adopted with minor amendments by the Group.¹⁴⁰ This text was then presented at the October 1987 Joint IAEA/NEA Working Group meeting, where it was further amended and, as a result of an internal reorganization of provisions, was finally adopted as Article III of the Joint Protocol.¹⁴¹ The text of Article III of the Joint Protocol thereafter remained unchanged until its final adoption in September 1988.

Despite the formulation of two conflict rules, Article III of the Joint Protocol achieves the same result that was originally envisaged when only one conflict rule was formulated. Having stated, in para. 1, that "either the Vienna Convention or the Paris Convention shall apply to a nuclear incident to the exclusion of the other", Article III, para. 2, relies on the principle of territoriality in order to lay down the conflict rule in the case of nuclear incidents occurring in nuclear installations (i.e. the place of the nuclear incident determines the applicable Convention), whereas Article III, para. 3, lays down the conflict rule in transport cases by simply referring to the Convention in force for the State in whose territory is situated the installation of the operator liable by virtue of the identical provisions of Article II.1(b)

- ¹³⁵ OECD-NEA document LEG/DOC(87)3 (Rev. 1), para. 39.
- ¹³⁶ IAEA document N5/TC/643 (4), para. 39.
- ¹³⁷ Article II of the 1974 Draft Joint Protocol reads as follows:

"For the purposes of this Protocol and taking account of the provisions of Article I thereof, either the Paris Convention or the Vienna Convention shall apply to a nuclear incident, to the exclusion of the other. The Convention applicable shall be that to which the Installation State of the operator liable by virtue of either Convention is a Party".

Having recalled the relevant paragraphs in WD II, the 1974 Explanatory Note explained that:

"This formula would also remove any doubt that in case of a nuclear incident involving only States Party to either the Paris Convention or the Vienna Convention, only that Convention would be applicable. The provisions of Article I [now Article IV of the Joint Protocol] would have to be taken into account only in those cases where Parties to the Paris Convention and to the Vienna Convention are involved" (para. 5).

¹³⁴ WD II, para. 54. See also Article 11 of the Paris Convention and Article VIII of the Vienna Convention. It must be pointed out, however, that the 1997 Vienna Protocol amends Article VIII of the Vienna Convention by adding a second paragraph providing for priority in the distribution of claims to claims in respect of loss of life or personal injury, in the event that the damage to be compensated exceeds the amount available under the Convention. No corresponding provision will be inserted in the Paris Convention by the 2004 amending Protocol.

These words were repeated almost verbatim in the 1986 Explanatory Note (para. 5). However, the words "for the purposes of this Protocol" were deleted at the June 1987 NEA Group meeting as a result of the adoption of the revised text prepared by the NEA Secretariat, which is referred to later in the text.

¹³⁸ 1987 Explanatory Text, para. 8.

¹³⁹ OECD-NEA document LEG/DOC(87)3 (Rev. 1), Annex IV.

¹⁴⁰ OECD-NEA document SEN/LEG(87)3, para. 35.

¹⁴¹ IAEA document N5/TC/643(3).

and (c) of the Vienna Convention and Article 4(a) and (b) of the Paris Convention. In both cases, the applicable Convention is the Convention in force for the Installation State. As is pointed out in Section 3.1, the idea that the operator remains liable under the Convention to which the State in which his installation is situated is a Party is now also made explicit in Article I of the Joint Protocol, which was first adopted in 1987 together with the revised text of Article III of the Joint Protocol.

A particular issue, which was discussed in 1987, relates to the applicable Convention in the case of different nuclear consignments —that is, where nuclear material is carried from or to an operator whose installation is situated in the territory of a Contracting Party to the Paris Convention and, at the same time and on the same means of transport (e.g. a ship), nuclear material is carried from or to another operator whose installation is situated in the territory of a Contracting Party to the Vienna Convention. In this respect, both the Note presented to the June 1987 NEA Group meeting and the "Background Material" presented to the October 1987 Joint IAEA/NEA Working Group meeting pointed out that:

"Which Convention applies is not a problem, where one of the operators has actually taken charge of the material or has accepted liability in writing.... [In that case,] that Convention will apply whose Contracting Party is the Installation State of the operator taking in charge. Where there is no actual taking in charge or no written acceptance of liability by one of the operators, the Convention applicable is only clear when the nuclear incident is caused exclusively by one of the nuclear consignments. Where it is caused by both consignments or — what is more likely, it is uncertain which one was responsible — both operators will be liable [Article 5(d) [of the Paris Convention], II.3 (a) [of the Vienna Convention]. Both Conventions are applicable, and the Protocol does not point to the exclusive application of one Convention. This legal position is however in no way the result of the Protocol, and would not be different without it. The advantage of the Protocol is precisely that it permits agreements between [Paris Convention and Vienna Convention] operators which exclude the simultaneous application of both Conventions."¹⁴²

¹⁴² OECD-NEA document LEG/DOC(87)3 (Rev. 1), para. 40, and IAEA document N5/TC/643 (4), para. 40.

4. FINAL CLAUSES AND TRANSITIONAL ISSUES

The original 1974 Draft Joint Protocol already contained a number of provisions (Articles III–IX) dealing with signature, ratification, accession, entry into force and denunciation of the treaty, as well as with depositary functions in respect thereof. These provisions were at the basis of the largely corresponding final clauses now contained in Articles V–XI of the Joint Protocol. These final clauses were provisionally adopted, on the basis of a draft prepared by the IAEA and NEA Secretariats, by the Joint IAEA/NEA Working Group in October 1987,¹⁴³ and thereafter remained unchanged, except for the date of the opening for signature of the Joint Protocol, which was inserted at the Diplomatic Conference in 1988. In addition to the issues already dealt with in the 1974 Draft Joint Protocol, the current final clauses also deal with denunciation, termination and authentic texts of the Joint Protocol. The final clauses do not envisage a specific amendment procedure, as it was felt that amendments could be dealt with in accordance with the general rules of the law of treaties as codified in Articles 39 and 41 of the 1969 Vienna Convention on the Law of Treaties.

4.1. DEPOSITARY FUNCTIONS AND AUTHENTIC TEXTS

Article VI.2 of the Joint Protocol designates the Director General of the IAEA as the depositary of the Joint Protocol and, consequently, as the recipient of instruments of ratification (acceptance, approval) or accession and of notifications by Contracting Parties. Under Articles VIII.1 and IX.1 of the Joint Protocol, notifications of denunciation and of termination of application of the Joint Protocol must also be directed to the depositary, whereas the usual depositary duties are specified in Articles X and XI of the Joint Protocol. The Secretary General of the OECD, which is the depositary of the Paris Convention, is mentioned in Articles X and XI as the beneficiary of those duties, together with the Contracting Parties to the Joint Protocol and the States invited to the 1988 Diplomatic Conference.

Under Article XI of the Joint Protocol, the Arabic, Chinese, English, French, Russian and Spanish texts of the Joint Protocol are equally authentic.

4.2. SIGNATURE, RATIFICATION, ACCESSION, ENTRY INTO FORCE, DENUNCIATION AND TERMINATION

As is pointed out in Section 2.1, the Joint Protocol is not intended as an amendment of either the Vienna Convention or the Paris Convention, both defined in Article I of the Joint Protocol so as to include "any amendment thereto which is in force for a Contracting Party", but aims at creating a "bridge" between the two Conventions in order to establish treaty relations between the Parties to one Convention that are not party to the other. Therefore, the Joint Protocol is not a free-standing international agreement and is only open to States which are already Parties to either the Vienna Convention or the Paris Convention, and only so long as they remain Parties thereto.

Thus, under Article V of the Joint Protocol, only States having signed, ratified or acceded to either the Vienna Convention or the Paris Convention were allowed to sign the Joint Protocol as from the date of its opening for signature, on 21 September 1988, until the date of its entry into force, on 27 April 1992. Under Article VI, the Joint Protocol is subject to ratification (acceptance or approval) on the part of the signatory States and is open to accession by non-signatory States, but instruments of ratification (acceptance, approval) or accession can only be accepted from States party to either the Vienna Convention or the Paris Convention. Conversely, under Article IX, the Joint Protocol automatically ceases to apply to any Contracting Party which ceases to be a Party to either the Vienna Convention or the Paris Convention.

¹⁴³ IAEA document N5/TC/643.

In accordance with Article VII.1, the Joint Protocol entered into force three months after the date of deposit of instruments of ratification (acceptance, approval) or accession "by at least five States Party to the Vienna Convention and five States Party to the Paris Convention", that is, on 27 April 1992. For each State ratifying (accepting, approving) or acceding to the Protocol after that date, the Protocol enters into force three months after the date of deposit of its instrument of ratification (acceptance, approval) or accession.

Under Article VIII, any Contracting Party may denounce the Joint Protocol by written notification to the depositary. In that case, the Joint Protocol ceases to apply to that Contracting Party one year after the date on which that notification is received. Moreover, as has been mentioned, under Article IX, the Joint Protocol automatically ceases to apply to a Contracting Party which has terminated application of either the Vienna Convention or the Paris Convention on the date when such termination takes effect, that is, quite irrespective of any formal denunciation of the Joint Protocol. Indeed, under Article IX of the Joint Protocol, such a Contracting Party is obliged to notify the depositary of the termination of the application of the relevant Convention with respect to it and of the date such termination takes effect.¹⁴⁴

In the event that, in accordance with Articles VIII or IX of the Joint Protocol, the number of the Contracting Parties to the Joint Protocol falls below the number necessary for its entry into force under Article VII.1, this does not automatically entail its termination. Under Article 55 of the 1969 Vienna Convention on the Law of Treaties, "unless the treaty otherwise provides, a multilateral treaty does not terminate by reason only of the fact that the number of the parties falls below the number necessary for its entry into force". Far from providing otherwise, the Joint Protocol, in Article VII.2, expressly provides that it "shall remain in force as long as both the Vienna Convention and the Paris Convention are in force", that is, quite irrespective of whether or not the number of the Parties thereto falls below the number necessary for its entry into force.

4.3. TRANSITIONAL ISSUES RELATING TO POTENTIAL CONFLICTS OF TREATY OBLIGATIONS

As far as the conditions for the entry into force of the Joint Protocol are concerned, the number of five States party to the Vienna Convention and five States party to the Paris Convention, which already appeared in the 1974 Draft Joint Protocol, was chosen because it corresponded to the number of ratifications or accessions required for the entry into force of both the Vienna Convention (Article XXIII) and the Paris Convention (Article 19(b)), and because it was considered to constitute an acceptable compromise between the interest of allowing the Joint Protocol to become effective within a reasonable period of time, on the one hand, and that of ensuring its practical application by a sufficient number of adhesions, on the other.¹⁴⁵

However, as is mentioned in Section 2.1, it was recognized at the same time that this compromise solution would not avoid "the creation of a 'third class' of countries"¹⁴⁶, that is, the Contracting Parties to the Joint Protocol as opposed to either the Contracting Parties to the Vienna Convention only or the Contracting Parties to the Paris Convention only. In other words, it was recognized from the very beginning that the smooth functioning of an instrument such as the Joint Protocol requires that as many Parties to either the Paris Convention or the Vienna Convention as possible (ideally all

¹⁴⁴ It may be of interest to recall in this respect that Article XX of the 1963 Vienna Convention provides that: "Notwithstanding the termination of the application of this Convention to any Contracting Party, either by termination pursuant to Article XXV or by denunciation pursuant to Article XXVI, the provisions of this Convention shall continue to apply to any nuclear damage caused by a nuclear incident occurring before such termination". Although Article XX is deleted in the Vienna Convention as amended by the 1997 Vienna Protocol (Article 18.1), a similar provision is contained in Article 22.4 of the Vienna Protocol and applies in case of denunciation of the 1997 Vienna Protocol (and thus of the 1997 Vienna Convention). There is no corresponding provision in the Paris Convention (in either its present form or as it will be amended by the 2004 Protocol), nor does the Joint Protocol itself contain any specific provision in this respect. Under the general rules of the law of treaties as codified in Article 70 of the 1969 Vienna Convention on the Law of Treaties, the termination of a treaty releases the Parties from any obligation to further perform the treaty but does not affect any right, obligation or legal situation of the Parties created through the execution of the treaty prior to its termination.

¹⁴⁵ 1974 Explanatory Note, para. 5.

¹⁴⁶ 1974 Explanatory Note, para. 5.

the Parties thereto) accept the treaty link offered by the Joint Protocol. In order to achieve that result, when the negotiations started again in the 1980s, consideration was given to the possibility of inserting in the Draft Joint Protocol the requirement that all future ratifications of, and accessions to, either the Vienna Convention or the Paris Convention must be accompanied by a ratification of, or accession to, the Joint Protocol, but that idea was discarded because it would have amounted to an amendment to both Conventions.¹⁴⁷ Alternatively, the NEA and IAEA Secretariats proposed that the number of ratifications required for the entry into force of the Joint Protocol be raised to two thirds of the Contracting Parties to the Vienna Convention and the Paris Convention, so as to make the Protocol attractive to future Contracting Parties of the Conventions as well.¹⁴⁸ However, when the final Draft Joint Protocol was adopted at the October 1987 Joint IAEA/NEA Working Group meeting, the original number was retained.

As for the kinds of problems that may transitionally arise for a Contracting Party to the Joint Protocol, so long as there are Contracting Parties to either the Paris Convention or the Vienna Convention that are not Party to the Joint Protocol, these are in essence similar to those that would have been caused by the ratification of the either the Vienna Convention or the Paris Convention by some (as opposed to all) States party to the other Convention (see Section 1.3). It was clearly pointed out from the early stages of the elaboration of the Joint Protocol that a limited number of ratifications or accessions "would not achieve in all cases the desired objective of removing simultaneous liability of two different operators and of ensuring that only one court is competent" in the event of a nuclear incident.¹⁴⁹

In particular, WD III contained more specific remarks on this issue and deserves full quotation. It was therein pointed out that:

"certain problems could arise in those cases where a nuclear incident would affect States Party to the Protocol and States Party to either Convention, but not to the Protocol. If, for example, an operator in a P-State which has not ratified the Protocol sends nuclear material to an operator in a V-State which has ratified the Protocol (Vp-State) through the territory of a P-State which has ratified the Protocol (Pp-State), and a nuclear incident occurs and damage is suffered in Pp, it might be argued that victims in Pp could bring actions for compensation against either the P-operator (pursuant to the Paris Convention) or the Vp-operator (pursuant to the Protocol in conjunction with the Vienna Convention), a choice which would not be possible without the Protocol being in force, as then only the Paris Convention would apply. It is to be noted that without the Protocol, victims in the transit P-State would not be protected under any Convention if the damage had been caused by a similar incident in V. If a Pp-operator sends nuclear material through the territory of P to a Vp-operator and a nuclear incident occurs in Vp which causes damage in P, an agreement between the Pp-operator and the Vp-operator that the latter should assume liability for the entire transport would not be binding as regards P. The Vp-operator could argue that the Vienna Convention would not be applicable as the nuclear damage was suffered in a non-Contracting State. The solution here (on which the transit P-State would insist) would be that the Pp-operator assumed liability for the transport. There are a number of variants to the above examples an analysis of which shows that the difficulties arising in the transitional period are by far less numerous and easier to resolve than those caused by the present situation where there is no relationship between the Conventions at all".¹⁵⁰

¹⁴⁷ OECD-NEA document LEG/DOC(87)3 (Rev. 1), para. 47.

¹⁴⁸ OECD-NEA document LEG/DOC(87)3 (Rev. 1), para. 48, and IAEA document N5/TC/643(4), para. 48.

¹⁴⁹ WD I, para. III.4.

¹⁵⁰ WD III, para. 17. "P" refers to the Paris Convention; "V" refers to the Vienna Convention.

JOINT PROTOCOL RELATING TO THE APPLICATION OF THE VIENNA CONVENTION AND THE PARIS CONVENTION

THE CONTRACTING PARTIES

HAVING REGARD to the Vienna Convention on Civil Liability for Nuclear Damage of 21 May 1963; HAVING REGARD to the Paris Convention on Third Party Liability in the Field of Nuclear Energy of 29 July 1960 as amended by the Additional Protocol of 28 January 1964 and by the Protocol of 16 November 1982;

CONSIDERING that the Vienna Convention and the Paris Convention are similar in substance and that no State is at present a Party to both Conventions;

CONVINCED that adherence to either Convention by Parties to the other Convention could lead to difficulties resulting from the simultaneous application of both Conventions to a nuclear incident; and

DESIROUS to establish a link between the Vienna Convention and the Paris Convention by mutually extending the benefit of the special regime of civil liability for nuclear damage set forth under each Convention and to eliminate conflicts arising from the simultaneous applications of both Conventions to a nuclear incident;

HAVE AGREED as follows:

ARTICLE I

In this Protocol:

- (a) "Vienna Convention" means the Vienna Convention on Civil Liability for Nuclear Damage of 21 May 1963 and any amendment thereto which is in force for a Contracting Party to this Protocol;
- (b) "Paris Convention" means the Paris Convention on Third Party Liability in the Field of Nuclear Energy of 29 July 1960 and any amendment thereto which is in force for a Contracting Party to this Protocol.

ARTICLE II

For the purpose of this Protocol:

- (a) The operator of a nuclear installation situated in the territory of a Party to the Vienna Convention shall be liable in accordance with that Convention for nuclear damage suffered in the territory of a Party to both the Paris Convention and this Protocol;
- (b) The operator of a nuclear installation situated in the territory of a Party to the Paris Convention shall be liable in accordance with that Convention for nuclear damage suffered in the territory of a Party to both the Vienna Convention and this Protocol.

ARTICLE III

1. Either the Vienna Convention or the Paris Convention shall apply to a nuclear incident to the exclusion of the other.

2. In the case of a nuclear incident occurring in a nuclear installation, the applicable Convention shall be that to which the State is a Party within whose territory that installation is situated.

3. In the case of a nuclear incident outside a nuclear installation and involving nuclear material in the course of carriage, the applicable Convention shall be that to which the State is a Party within whose territory the nuclear installation is situated whose operator is liable pursuant to either Article II.1(b) and (c) of the Vienna Convention or Article 4(a) and (b) of the Paris Convention.

ARTICLE IV

1. Articles I to XV of the Vienna Convention shall be applied, with respect to the Contracting Parties to this Protocol which are Parties to the Paris Convention, in the same manner as between Parties to the Vienna Convention.

2. Articles 1 to 14 of the Paris Convention shall be applied, with respect to the Contracting Parties to this Protocol which are Parties to the Vienna Convention, in the same manner as between Parties to the Paris Convention.

ARTICLE V

This Protocol shall be open for signature, from 21 September 1988 until the date of its entry into force, at the Headquarters of the International Atomic Energy Agency by all States which have signed, ratified or acceded to either the Vienna Convention or the Paris Convention.

ARTICLE VI

1. This Protocol is subject to ratification, acceptance, approval or accession. Instruments of ratification, acceptance or approval shall only be accepted from States Party to either the Vienna Convention or the Paris Convention. Any such State which has not signed this Protocol may accede to it.

2. The instruments of ratification, acceptance, approval or accession shall be deposited with the Director General of the International Atomic Energy Agency, who is hereby designated as the depositary of this Protocol.

ARTICLE VII

1. This Protocol shall come into force three months after the date of deposit of instruments of ratification, acceptance, approval or accession by at least five States Party to the Vienna Convention and five States Party to the Paris Convention. For each State ratifying, accepting, approving or acceding to this Protocol after the deposit of the above-mentioned instruments this Protocol shall enter into force three months after the date of deposit of the instrument of ratification, acceptance, approval or accession.

2. This Protocol shall remain in force as long as both the Vienna Convention and the Paris Convention are in force.

ARTICLE VIII

1. Any Contracting Party may denounce this Protocol by written notification to the depositary.

2. Denunciation shall take effect one year after the date on which the notification is received by the depositary.

ARTICLE IX

1. Any Contracting Party which ceases to be a Party to either the Vienna Convention or the Paris Convention shall notify the depositary of the termination of the application of that Convention with respect to it and of the date such termination takes effect.

2. This Protocol shall cease to apply to a Contracting Party which has terminated application of either the Vienna Convention or the Paris Convention on the date such termination takes effect.

ARTICLE X

The depositary shall promptly notify Contracting Parties and States invited to the Conference on the relationship between the Paris Convention and the Vienna Convention as well as the Secretary General of the Organisation for Economic Co-operation and Development of:

- (a) Each signature of this Protocol;
- (b) Each deposit of an instrument of ratification, acceptance, approval or accession concerning this Protocol;
- (c) The entry into force of this Protocol;
- (d) Any denunciation; and
- (e) Any information received pursuant to Article IX.

ARTICLE XI

The original of this Protocol, of which the Arabic, Chinese, English, French, Russian and Spanish texts are equally authentic, shall be deposited with the depositary, who shall send certified copies to Contracting Parties and States invited to the Conference on the relationship between the Paris Convention and the Vienna Convention as well as the Secretary General of the Organisation for Economic Co-operation and Development.

IN WITNESS WHEREOF the undersigned being duly authorized by their respective Governments for that purpose have signed the present Joint Protocol.

DONE at Vienna this twenty-first day of September, one thousand nine hundred and eighty-eight.



Where to order IAEA publications

In the following countries IAEA publications may be purchased from the sources listed below, or from major local booksellers. Payment may be made in local currency or with UNESCO coupons.

AUSTRALIA

DA Information Services, 648 Whitehorse Road, MITCHAM 3132 Telephone: +61 3 9210 7777 • Fax: +61 3 9210 7788 Email: service@dadirect.com.au • Web site: http://www.dadirect.com.au

BELGIUM

Jean de Lannoy, avenue du Roi 202, B-1190 Brussels Telephone: +32 2 538 43 08 • Fax: +32 2 538 08 41 Email: jean.de.lannoy@infoboard.be • Web site: http://www.jean-de-lannoy.be

CANADA

Bernan Associates, 4501 Forbes Blvd, Suite 200, Lanham, MD 20706-4346, USA Telephone: 1-800-865-3457 • Fax: 1-800-865-3450 Email: customercare@bernan.com • Web site: http://www.bernan.com

Renouf Publishing Company Ltd., 1-5369 Canotek Rd., Ottawa, Ontario, K1J 9J3 Telephone: +613 745 2665 • Fax: +613 745 7660 Email: order.dept@renoufbooks.com • Web site: http://www.renoufbooks.com

CHINA

IAEA Publications in Chinese: China Nuclear Energy Industry Corporation, Translation Section, P.O. Box 2103, Beijing

CZECH REPUBLIC

Suweco CZ, S.R.O., Klecakova 347, 180 21 Praha 9 Telephone: +420 26603 5364 • Fax: +420 28482 1646 Email: nakup@suweco.cz • Web site: http://www.suweco.cz

FINLAND

Akateeminen Kirjakauppa, PO BOX 128 (Keskuskatu 1), FIN-00101 Helsinki Telephone: +358 9 121 41 • Fax: +358 9 121 4450 Email: akatilaus@akateeminen.com • Web site: http://www.akateeminen.com

FRANCE

Form-Edit, 5, rue Janssen, P.O. Box 25, F-75921 Paris Cedex 19 Telephone: +33 1 42 01 49 49 • Fax: +33 1 42 01 90 90 Email: formedit@formedit.fr • Web site: http://www. formedit.fr

Lavoisier SAS, 145 rue de Provigny, 94236 Cachan Cedex Telephone: + 33 1 47 40 67 02 • Fax +33 1 47 40 67 02 Email: romuald.verrier@lavoisier.fr • Web site: http://www.lavoisier.fr

GERMANY

UNO-Verlag, Vertriebs- und Verlags GmbH, Am Hofgarten 10, D-53113 Bonn Telephone: + 49 228 94 90 20 • Fax: +49 228 94 90 20 or +49 228 94 90 222 Email: bestellung@uno-verlag.de • Web site: http://www.uno-verlag.de

HUNGARY

Librotrade Ltd., Book Import, P.O. Box 126, H-1656 Budapest Telephone: +36 1 257 7777 • Fax: +36 1 257 7472 • Email: books@librotrade.hu

INDIA

Allied Publishers Group, 1st Floor, Dubash House, 15, J. N. Heredia Marg, Ballard Estate, Mumbai 400 001, Telephone: +91 22 22617926/27 • Fax: +91 22 22617928 Email: alliedpl@vsnl.com • Web site: http://www.alliedpublishers.com

Bookwell, 2/72, Nirankari Colony, Delhi 110009 Telephone: +91 11 23268786, +91 11 23257264 • Fax: +91 11 23281315 Email: bookwell@vsnl.net

ITALY

Libreria Scientifica Dott. Lucio di Biasio "AEIOU", Via Coronelli 6, I-20146 Milan Telephone: +39 02 48 95 45 52 or 48 95 45 62 • Fax: +39 02 48 95 45 48 Email: info@libreriaaeiou.eu

JAPAN

Maruzen Company Ltd, 1-9-18, Kaigan, Minato-ku, Tokyo, 105-0022 Telephone: +81 3 6367 6079 • Fax: +81 3 6367 6207 Email: journal@maruzen.co.jp • Web site: http://www.maruzen.co.jp

REPUBLIC OF KOREA

KINS Inc., Information Business Dept. Samho Bldg. 2nd Floor, 275-1 Yang Jae-dong SeoCho-G, Seoul 137-130 Telephone: +02 589 1740 • Fax: +02 589 1746 • Web site: http://www.kins.re.kr

NETHERLANDS

De Lindeboom Internationale Publicaties B.V., M.A. de Ruyterstraat 20A, NL-7482 BZ Haaksbergen Telephone: +31 (0) 53 5740004 • Fax: +31 (0) 53 5729296 Email: books@delindeboom.com • Web site: http://www.delindeboom.com

Martinus Nijhoff International, Koraalrood 50, P.O. Box 1853, 2700 CZ Zoetermeer Telephone: +31 793 684 400 • Fax: +31 793 615 698 Email: info@nijhoff.nl • Web site: http://www.nijhoff.nl

Swets and Zeitlinger b.v., P.O. Box 830, 2160 SZ Lisse Telephone: +31 252 435 111 • Fax: +31 252 415 888 Email: infoho@swets.nl • Web site: http://www.swets.nl

NEW ZEALAND

DA Information Services, 648 Whitehorse Road, MITCHAM 3132, Australia Telephone: +61 3 9210 7777 • Fax: +61 3 9210 7788 Email: service@dadirect.com.au • Web site: http://www.dadirect.com.au

SLOVENIA

Cankarjeva Zalozba d.d., Kopitarjeva 2, SI-1512 Ljubljana Telephone: +386 1 432 31 44 • Fax: +386 1 230 14 35 Email: import.books@cankarjeva-z.si • Web site: http://www.cankarjeva-z.si/uvoz

SPAIN

Díaz de Santos, S.A., c/ Juan Bravo, 3A, E-28006 Madrid Telephone: +34 91 781 94 80 • Fax: +34 91 575 55 63 Email: compras@diazdesantos.es, carmela@diazdesantos.es, barcelona@diazdesantos.es, julio@diazdesantos.es Web site: http://www.diazdesantos.es

UNITED KINGDOM

The Stationery Office Ltd, International Sales Agency, PO Box 29, Norwich, NR3 1 GN Telephone (orders): +44 870 600 5552 • (enquiries): +44 207 873 8372 • Fax: +44 207 873 8203 Email (orders): book.orders@tso.co.uk • (enquiries): book.enquiries@tso.co.uk • Web site: http://www.tso.co.uk

On-line orders DELTA Int. Book Wholesalers Ltd., 39 Alexandra Road, Addlestone, Surrey, KT15 2PQ Email: info@profbooks.com • Web site: http://www.profbooks.com

Books on the Environment Earthprint Ltd., P.O. Box 119, Stevenage SG1 4TP Telephone: +44 1438748111 • Fax: +44 1438748844 Email: orders@earthprint.com • Web site: http://www.earthprint.com

UNITED NATIONS

Dept. 1004, Room DC2-0853, First Avenue at 46th Street, New York, N.Y. 10017, USA (UN) Telephone: +800 253-9646 or +212 963-8302 • Fax: +212 963-3489 Email: publications@un.org • Web site: http://www.un.org

UNITED STATES OF AMERICA

Bernan Associates, 4501 Forbes Blvd., Suite 200, Lanham, MD 20706-4346 Telephone: 1-800-865-3457 • Fax: 1-800-865-3450 Email: customercare@bernan.com · Web site: http://www.bernan.com

Renouf Publishing Company Ltd., 812 Proctor Ave., Ogdensburg, NY, 13669 Telephone: +888 551 7470 (toll-free) • Fax: +888 568 8546 (toll-free) Email: order.dept@renoufbooks.com • Web site: http://www.renoufbooks.com

Orders and requests for information may also be addressed directly to:

Marketing and Sales Unit, International Atomic Energy Agency

Vienna International Centre, PO Box 100, 1400 Vienna, Austria Telephone: +43 1 2600 22529 (or 22530) • Fax: +43 1 2600 29302 Email: sales.publications@iaea.org • Web site: http://www.iaea.org/books

The Joint Protocol Relating to the Application of the Vienna Convention and the Paris Convention (the Joint Protocol) was adopted in September 1988 by the Conference on the Relationship between the Paris Convention and the Vienna Convention, which met in Vienna under the joint auspices of the IAEA and the OECD Nuclear Energy Agency. This publication reproduces the text of the Joint Protocol together with an explanatory text which was prepared and finalized by the International Expert Group on Nuclear liability (INLEX). The explanatory text on the Joint Protocol complements the explanatory texts on the 1997 Vienna Convention on Civil Liability for Nuclear Damage and the 1997 Convention on Supplementary Compensation for Nuclear Damage, which were previously developed by INLEX and published as IAEA International Law Series No. 3.

> INTERNATIONAL ATOMIC ENERGY AGENCY VIENNA ISBN 978-92-0-139410-1 ISSN 1991-2366